Basel Compliance and Financial Stability: Evidence from Islamic Banks



Mohammad Bitar¹ · Sami Ben Naceur² · Rym Ayadi³ · Thomas Walker⁴

Received: 18 October 2018 / Revised: 24 May 2020 / Accepted: 28 May 2020 / Published online: 15 September 2020 © The Author(s) 2020

Abstract

We find that compliance with the Basel Core Principles (BCPs) has a strong positive effect on the stability of conventional banks, and a positive but less pronounced effect on the stability of Islamic banks. We also find that the main impact of compliance is an increase in capital ratios, whereas other components of the Z-score are negatively affected. This reflects the desire of banks to be more closely integrated into the global financial system by holding higher capital ratios. The findings also justify the 2015 decision of the Islamic Financial Services Board to publish similar principles for Islamic banks.

JEL Classification Numbers G18 · G21 · P51

Keywords BCPs · Core principles for Islamic finance regulation · Stability · Islamic banks

The views expressed here are those of the authors, and the paper does not represent the views of the IMF, its Executive Board or its management.

Mohammad Bitar mohammad.bitar@nottingham.ac.uk

> Sami Ben Naceur sbennaceur@imf.org

Rym Ayadi rym.ayadi@city.ac.uk

Thomas Walker thomas.walker@concordia.ca

- ¹ Nottingham University, Nottingham, UK
- ² International Monetary Fund, Washington, D.C., USA
- ³ Cass Business School, London, UK
- ⁴ Concordia University in Canada, Montreal, Canada

1 Introduction

In this study, we examine whether compliance with the Basel Core Principles (BCPs) for effective banking supervision affects the stability and risk taking of conventional and Islamic banks. In prior studies, Demirgüç-Kunt and Detragiache (2011) and Ayadi et al. (2016) have examined the effects of BCP compliance using large and heterogeneous samples of conventional banks around the world. In this paper we extend their analyses to include Islamic banks. We also focus on a more homogeneous sample in the sense that the banks in our study operate mainly in developing and emerging countries.

The BCPs were introduced in 1997 by the Basel Committee on Banking and Supervision (BCBS). Since then, several surveys have been conducted by the International Monetary Fund (IMF) and the World Bank to assess the quality of banking regulation and supervision worldwide under the Basel principles. The principles were initially created as a pilot project for 12 advanced countries, but they rapidly became the global standard for banking regulation. BCPs include 25 principles organised into 7 chapters (i.e., each chapter contains a few of the 25 principles). However, one important drawback of the BCPs is that they do not take into account the distinctive characteristics of certain types of bank such as Islamic banks.¹

In 2015, the Islamic Financial Services Board (IFSB),² an international regulatory organization that was created to promote the development and the stability of the Islamic financial industry, published a set of guidelines called the Core Principles for Islamic Finance Regulation (CPIFRs). These guidelines are built on BCBS standards and have been extended to deal with the characteristics of Islamic banks.

Within these guidelines, some of the CPIFRs replicate and fully reflect the Basel principles, some represent amendments to the BCPs, and some CPIFRs are completely new. Because the CPIFR guidelines were published in 2015, Islamic banks were not expected to implement them before January 2016 and in many cases even later (IFSB 2015). Thus, data on Islamic bank compliance with the CPIFRs are not available at this stage. However, since some of the CPIFRs are similar to conventional bank BCPs, and there are data available on the compliance of Islamic banks with those principles, our study examines whether the adoption of the BCPs has affected the stability of Islamic banks. This enables us to make some plausible predictions about the expected effects of the CPIFRs on the financial soundness of Islamic banks.

To examine these issues, we use a sample of 761 conventional and Islamic banks operating in 19 countries and covering the period from 1999 to 2013. In contrast to Demirgüç-Kunt and Detragiache (2011), our findings suggest that BCP compliance is positively associated with the stability of conventional banks (at the 1 % significance level) and that this holds true for at least five of the seven

¹ Islamic banks are by nature financial intermediaries that are compliant with *Sharia'a* law (Gheeraert 2014). Thus, they can be defined as institutions that allocate resources and invest them under the guidance of *Sharia'a* principles without the use of interest. Islamic banks operate in a highly regulated industry. However, due to the special characteristics of Islamic banks—i.e. the concept of profit and loss sharing at both the asset side (with entrepreneurs/borrowers) and the liability side (with depositors/investors)—they not only adhere to the Basel Committee on Banking and Supervision (BCBS) regulatory guidelines but also to specific capital guidelines by the Islamic Financial Services Board (IFSB) and the Accounting and Auditing Organization for Islamic Financial Institutions (AAOIFI). In this paper, we do not detail the characteristics of Islamic banks because they were already reviewed extensively in the previous literature. For excellent reviews, the reader may refer to Khan (2010), Beck et al. (2013), and Abedifar et al. (2013).

² Established in Kuala Lumpur, Malaysia, in 2002, the Islamic Financial Services Board (IFSB) comprises 188 members, including 61 regulatory authorities, 8 intergovernmental organizations, and 119 market players. The IFSB is considered to be the complement of the Basel Committee on Banking Supervision.

chapters that make up the BCP. The effect remains positive but less pronounced for Islamic banks, where the effects of compliance are positive and significant at the 5 % level or better for three of the seven chapters. A deeper examination of the components of the main dependent variable (bank Z-scores) shows that the component of stability most influenced by compliance is the bank capital ratio. The findings indicate that adherence to international regulatory standards improves the stability of the two bank types through incentives to hold higher capital ratios. The results hold when controlling for bank financial characteristics, as well as for the macroeconomic and institutional environment. The findings also remain invariant (i) across sub-samples representing different regions, economic cycles, market disciplines, and political and institutional environments, (ii) when employing alternative measures of risk and stability, (iii) when using an instrumental variables approach and the Heckman estimation technique to address potential endogeneity and selection biases, and (iv) when using propensity score matching (PSM) to reduce any biases due to the limited sample size.

This study contributes to the literature on both conventional and Islamic banks in at least three important ways. First, we highlight a strong positive impact of the BCP index on the stability of conventional banks, while the effect remains positive but somewhat weaker for Islamic banks. This provides regulatory organizations such as the Bank for International Settlements (BIS) and the IFSB with initial empirical evidence to support the effective role of BCP standards in improving bank stability. Given that the BCPs are also effective in improving the stability of Islamic banks, it is likely that the CPFIRs will positively affect the stability of those banks, as the CPFIRs are modelled closely on the BCPs.

Second, we show that regulatory compliance enhances bank stability through two main mechanisms: (i) the tendency to make prudent investment decisions, which avoid risky activities and thereby result in lower but more stable returns on assets; and (ii) the maintenance of strong insolvency ratios driven by the desire of banks in developing countries to be recognized and better integrated into the global financial system.

Finally, we add to the comparative literature on conventional and Islamic banks (Abedifar et al. 2013; Mollah et al. 2017; Bitar et al. 2017; Bitar and Tarazi 2019) by exploring the regulatory determinants of bank stability and by documenting compelling evidence that the two bank types respond to similar determinants.

The remainder of this paper is structured as follows: Section II briefly reviews the literature. Section III describes the sample, outlines our empirical approach, and provides variable definitions. Section IV presents the main results, while Section V examines the robustness of the findings. The last section provides a brief summary and comments on future directions for the research.

2 Literature Review

The literature examining the effect of banking regulation on the risks to, and the stability of, the financial system does not employ a consistent set of indicators that can be used as a proxy for banking regulation. While some studies make use of accounting and market ratios such as regulatory capital, liquidity, and leverage measures, others use proxies based on questionnaires and surveys administered by governments and international regulatory organizations. These studies often report inconclusive and contradictory results.

Barth et al. (2004, 2006, and 2008) were among the first to examine the effect of banking regulation and supervision on bank performance and stability using international data. Their

findings suggest that strong monitoring of markets and the private sector is an important factor in promoting the performance and stability of the financial sector. Focusing on corporate governance, Leaven and Levine (2009) examine the effects of banking regulation and supervision using a variety of proxies (capital requirements, capital stringency, activity restrictions, and deposit insurance), while also considering bank ownership structure. They conclude that regulation increases bank risk-taking when a bank has an ultimate owner, while the opposite occurs when a bank is widely held. Klomp and de Haan (2012) investigate whether banking regulation has a homogeneous effect on bank risk. Using a sample of 200 banks from 21 OECD countries, their findings show that banking regulation is more effective in improving safety for riskier banks; i.e., the effect of regulation is not uniform and depends on a bank's risk profile. Klomp and de Haan (2014) extend this investigation by examining whether the association between banking regulation and risk varies with the level of development of a country's institutional environment. Using a sample of 400 banks from 70 developing and emerging countries, their findings indicate that the positive effect of banking regulation and supervision on risk avoidance is enhanced in countries with a better institutional environment.

More recently, Doumpos et al. (2015) use a large sample of 1700 commercial banks operating in 90 countries over the period 2000–2011 to study the effect of three indices of regulation (central bank independence, central bank involvement in prudential regulation, and supervisory unification) on bank stability. Depending on bank size and the country's official supervisory power, their results yield a positive and significant association with bank Z-scores, especially during periods of crisis. Finally, using a sample of 19 EU countries covering the 1999–2011 period, Carretta et al. (2015) investigate whether the culture of banking supervision (proxied by Hofstede's cultural dimensions) affects the stability of banks. Their findings suggest that so-called supervisory cultures, i.e. cultures based on collectivism and avoidance of uncertainty, are positively correlated with bank Z-scores. Accordingly, they highlight the importance of cultural dimensions when evaluating the success of banking regulation by the European Central Bank (ECB).

A serious shortcoming of these studies is that they evaluate the effectiveness of banking regulation and supervision based on what is claimed in the banks' books rather than on an objective assessment of whether, and to what extent, these regulations are actually implemented (Demirgüç-Kunt and Detragiache 2011; Ayadi et al. 2016). Such an assessment often reveals a lack of proper implementation, especially in developing countries, so that the variation across different countries according to the banks' books may underestimate the true variation in the execution of banking regulations (Demirgüç-Kunt and Detragiache 2011).

A small but growing stream of literature utilizes the BCP index to assess the quality of banking supervision as an alternative to measures based on questionnaires and surveys (as reported above). Developed by the World Bank and the IMF under the Basel Core Financial Sector Assessment Program (FSAP), the BCP index is considered a unique source of information that represents the quality of supervision and regulation in countries around the globe. Demirgüç-Kunt and Detragiache (2011) argue that assessments by the FSAP are reliable for two reasons: First, the BCP index reflects the actual implementation of the different elements of banking regulation and supervision. Second, assessments are based on an explicit and standardized methodology and are conducted by experienced international assessors with broad country experience.³

³ However, BCP assessments cannot be considered an exact science, as the evaluations might be affected by several factors depending on the assessors' subjectivity and experience as well as the existing regulatory framework (Demirgüç-Kunt and Detragiache 2011; Ayadi et al. 2016).

Several studies have employed the BCP index as a proxy measure of regulatory compliance and have used it to examine the effect of compliance on the performance and stability of the banking system. Sundararajan et al. (2001) examine the association between BCP compliance and bank soundness, using a sample of banks in 25 countries. Their findings highlight the importance of other bank-level and macroeconomic factors and demonstrate that the implementation of international standards is not sufficient in itself to ensure financial soundness. Demirgüç-Kunt et al. (2008) find that countries that comply with the BCP requirements for information disclosure receive higher financial strength ratings from Moody's. In addition, Podpiera (2006) investigates the effect of BCP compliance on bank performance using a sample of banks from advanced, emerging, and developing countries. He finds that banks in countries with higher compliance have lower levels of non-performing loans and lower net interest margins. In a similar study, Cihak and Tieman (2008) show that BCP compliance is positively and strongly associated with a country's level of governance and GDP per capita, while the effect is less significant when replacing the BCP index with on-the-book regulatory measures. Demirgüç-Kunt and Detragiache (2011) investigate the association between BCP compliance and financial stability for a sample of international banks. Employing an overall compliance index, the authors find no evidence of a significant relationship between the index and the banks' Z-scores. Finally, Ayadi et al. (2016) extend the work of Demirgüç-Kunt and Detragiache (2011) by focusing on bank efficiency. Their results show no association between BCP compliance and efficiency. Table 1 summarises the available literature on BCP studies.

Given that the BCPs are designed to promote the stability and financial soundness of conventional banks, it might be argued that the likelihood of BCP compliance having an effect on the stability of Islamic banks would be negligible or, at best, relatively slim. This conclusion is predicated on the assumption that Islamic banks have different balance sheets and different financial products from conventional banks. However, the literature offers a variety of opinions on whether or not, in practice, Islamic banks share the same financial characteristics as conventional banks. The main reason for the differences of opinion is that the current business models of Islamic banks are inconsistent, such that in some cases there is a significant disjunction between Sharia'a ideals and bank practices (Khan 2010). One would expect that under Sharia'a law, Profit Loss Sharing (PLS) instruments such as Musharaka and Moudharaba (core elements of Islamic banking and finance) would dominate Islamic banking practices. Yet, surprisingly, non-PLS mark-up modes of finance, such as *Murabaha* and *Ijara*, predominate. Mark-up financing techniques are considered less Sharia'a compliant and are seen as the hallmark of conventional banking; hence their predominance in the Islamic sector suggests that the two bank types may in fact be quite similar (Abedifar et al. 2013; Beck et al. 2013; Bitar and Tarazi 2019).

Recently, the IFSB published new guidelines, the CPIFRs (IFSB 2015), which are based on the Basel Core Principles for effective banking supervision. According to the IFSB, the proposed guidelines aim to "build on the standards adopted by relevant conventional standards [...] and to adapt or supplement them only to the extent necessary to deal with the specificities of Islamic finance" (IFSB 2015, p. 2). A detailed description of the CPIFRs is presented in Table 12.

CPIFR guidelines differ from BCP guidelines in at least three main respects. First, in the CPIFR framework, investment account holders (IAHs) are treated more like investors than depositors. This has an impact on capital adequacy ratios, the results of the relevant risk weighting methodology, the role of regulatory authorities in capital treatment, the bank's smoothing behavior, and the bank's exposure to displaced commercial risk. Second, the rate of

Table 1 Overview of Basel	Core Pri	Table 1 Overview of Basel Core Principles studies in conventional banking. This table presents a summary of selected literature on the Basel Core Principles	This table press	ents a summary of selected literature or	n the Basel Core Principles
Authors	Year	Countries in sample	Sample period	Dependent variable(s)	Effect of BCPs (sign)
Podpiera	2006	65 countries (13 advanced, 19 emerging, and 33 developing countries)	1998–2002 1998–2001	1998–2002 Non-performing loans 1998–2001 Net interest margin	Û.
Sundararajan et al.	2001	35 countries	1999–2000	Spread risk Non-performing loans	insignificant insignificant
Demirgüç-Kunt et al.	2008	39 countries	1999–2003	Moody's ratings of bank financial strength	(+) Basel core principle related to information disclosure (i.e., Chapter 5 Information Requirements)
Cihak and Tieman	2008	not disclosed	not disclosed	Implementation of BCPs	(-) Non-performing loans (+) GDP ner canita
Demirgüç-Kunt and Detragiache	2011	86 countries	1999–2006 Z-score	Z-score	insignificant
Ayadi et al.	2016	75 countries	1999–2014	1999–2014 DEA efficiency scores	insignificant

return (ROR) risk differs in that it depends on market conditions and on competition with conventional banks. The ROR might lead to the use of bank reserves or to displaced commercial risk (DCR) if an Islamic bank absorbs any losses (partially or entirely), if reducing its share of profits yields a shortcoming in the returns payable to IAHs, or through a donation from the shareholders' share of income. Finally, the CPIFRs stipulate that Islamic banks must possess an effective *Sharia'a* governance system to ensure the compliance of the banks' activities, investments and products with Islamic law; obviously, the BCP guidelines do not include this requirement.

As mentioned above, differences between Islamic and conventional banks might influence the extent to which BCP compliance affects the stability of Islamic banks compared to their conventional counterparts. While CPIFRs take these differences into account, an empirical investigation of the effect of CPIFRs on the stability of Islamic banks is not possible at this stage because CPIFRs have only been introduced recently. We thus substitute compliance with the BCPs (measured by the BCP index) for CPIFR compliance, and we argue that the BCPs should have a similar effect to the CPIFRs for two reasons: First, according to the IFSB (2015), seven principles in the CPIFR guidelines were kept the same as the BCPs, seventeen principles were amended, and only one principle was replaced. The main difference resides in four new CPIFR principles related to the characteristics of Islamic banks that were not considered in the BCP guidelines. Second, given that CPIFR guidelines resemble BCP guidelines, they are likely to have similar effects on banks with similar characteristics - and the literature suggests that the two bank types are indeed fairly similar (or at least not very different) in terms of business orientation (Beck et al. 2013), stability and interest (financing) margins (Abedifar et al. 2013), profitability (Mollah and Zaman 2015), and capital structure (Bitar and Tarazi 2019).

3 Data and Methodology

3.1 Sample Construction

In order to investigate the effect of the Basel Core Principles on the stability and risk of conventional and Islamic banks, we compile data from three main sources: (1) the IMF and the World Bank Basel Core Financial Sector Assessment Program (FSAP) database,⁴ which contains detailed information on country evaluation of, and compliance with, the BCPs during the period 1999–2012; (2) the World Bank's World Development Indicators (WDI) and World Governance Indicators (WGI) for macro-economic and governance variables; and (3) the Bankscope Database provided by Bureau van Dijk and Fitch Ratings for yearly-based accounting data.

In the selection of bank-level data, we follow the comparative literature on conventional and Islamic banks (Abedifar et al. 2013; Beck et al. 2013; Bitar and Tarazi 2019) and retrieve financial information from 1999 to 2013 in 33 countries where both bank types exist. A bank is excluded from the sample if it does not have at least three continuous observations. In addition, we remove countries that have data for fewer than 4 banks (Beck et al. 2013). In contrast to Ayadi et al. (2016), our study focuses on a broad sample of listed and unlisted banks—not just

⁴ Some assessments are publicly available through the IMF and World Bank websites. The remainder are kept confidential by the countries' authorities. In this study, we use a private database provided by the IMF which contains assessments for all countries. We thank the IMF for sharing these data.

publicly listed banks—to ensure an adequate representation of Islamic banks, most of which are unlisted.

We then match bank-level information with country-level information to control for variations in a country's macroeconomic and regulatory conditions. After checking the FSAP database, we find 28 countries that provide information on their compliance with the BCPs and that are home to both types of bank. We exclude countries such as Algeria, Bosnia, Brunei, the Cayman Islands, Iran, Iraq, Qatar, Senegal, the Sudan, and Yemen because of missing information on some of the BCP chapters. Our final sample is thus reduced to 651 conventional banks and 110 Islamic banks operating in 19 countries⁵ and is characterized by the homogeneity that results from these countries having similar financial characteristics and macroeconomic conditions. Some of these countries have few Islamic banks while others have many Islamic banks.

Because the IMF and the World Bank collected data on BCP chapter compliance in three different waves (1999, 2005 and 2012), our analyses are carried out as follows: (i) the 1999 data are used for the period 1999–2004, (ii) the 2005 data are used for the period 2005–2011, and (iii) the 2012 data are used for the period 2012–2013. In our sample, countries witnessed one or two assessment points. For instance, Pakistan only reported its BCP compliance in 2004. Thus, to avoid losing observations, the 2004 data are used for the period 2005–2013. Similarly, Saudi Arabia reported its BCP compliance in 2004 and 2011. Accordingly, we use the 2004 data for the period 2004–2010 and the 2011 data for the period 2011–2013.

3.2 Empirical Approach and Definition of Variables

The main dependent variable used to evaluate bank stability is the Z-score and the main independent variable is the country's BCP compliance index. We follow Mollah and Zaman (2015) and Bitar et al. (2016) and use random-effect GLS regressions to examine the effect of BCP compliance on bank financial stability. We prefer the GLS technique over other estimation techniques for two reasons: First, regression models such as OLS ignore the panel structure of our data. Second, our Islamic bank dummy is time-invariant and cannot be estimated using a fixed-effects methodology. Accordingly, we use the following baseline regression equations:

$$\begin{aligned} Stability_{ijt} &= \alpha + \beta_1 \times BCP_{jt} + \beta_2 \times Bank_control_{ijt-1} + \beta_3 \times Country_control_{jt} + \sum_{t=1}^{T} \mu_t \\ &\times Time_t + \varepsilon_{ijt} \end{aligned}$$
(1)

$$\begin{aligned} Stability_{ijt} &= \alpha + \beta_1 \times Islamic_i + \beta_2 \times BCP_{jt} + \beta_3 \times BCP_{jt} \times Islamic_i + \beta_4 \\ &\times Bank_control_{ijt-1} + \beta_5 \times Country_control_{jt} + \sum_{t=1}^{T} \mu_t \times Time_t + \varepsilon_{ijt} \end{aligned}$$
(2)

where *Stability*_{*ijt*} represents the natural logarithm of the Z-score (defined below) of bank *i* in country *j* at time *t*, and BCP_{jt} denotes the Basel Core Principles compliance index for country *j* at time *t* (if a country has reported its BCP compliance at least once). *Bank*_control_{*ijt*-1} is a

⁵ The sample is dominated by developing countries. Two developed countries are also included: The UK and Singapore. We include these countries because they have available data on BCP chapters and are markets where both conventional and Islamic banks operate side by side.

vector of bank-level control variables. Country _control_{jt} is a vector of country-level control variables. Time_t represents year fixed effects while ε_{ijt} is a random disturbance, assumed to be normally distributed with zero mean and constant variance, ε_{ijt} -*iid* $N(0, \sigma^2)$. In Equation (2), Islamic_i is a dummy variable taking on a value of one for Islamic banks and zero for conventional banks. Finally, an interaction term is introduced between Islamic and BCP to investigate whether a country's compliance with the BCP affects the stability of Islamic banks differently from the stability of their conventional counterparts.

The Z-score is defined as ([return on average assets + equity/assets]/[standard deviation of the return on average assets]) over the period (t, t–3) (Groeneveld and de Vries 2009; Beck et al. 2013; Abedifar et al. 2013; Mollah et al. 2017). In our regression analysis, we use the natural logarithm of the Z-score to minimize the effects of higher values, which could represent outliers (Carretta et al. 2015). In our robustness tests, we follow Bitar et al. (2017) and use loan-loss reserves to gross loans (LLRGL), loan-loss provisions to total loans (LLPTL), nonperforming loans to gross loans (NPLGL), and the volatility of the net-interest margin (SDNIM) to examine the impact of the BCP compliance index on the stability and risk of the two bank types.

Despite the plethora of research on banking regulation and supervision, the BCP compliance index is rarely used in studies of conventional banking and, to the best of our knowledge, has never been used in an Islamic banking context. The BCP index is based on 25 principles that are considered the best measures of compliance with banking regulation and supervision. These principles are classified into seven chapters as follows: (Ch. 1) Preconditions for Effective Banking Supervision; (Ch. 2) Licensing and Structure; (Ch. 3) Prudential Regulation and Requirements; (Ch. 4) Methods of Ongoing Supervision; (Ch. 5) Information Requirements; (Ch. 6) Formal Powers of Supervisors; and (Ch. 7) Cross-Border Banking. Definitions of the different principles used to construct the seven chapters are provided in Table 13.

We use both an aggregate approach to examine the effect of the overall BCP index and a disaggregate approach, in which the effect of compliance with individual chapters on bank stability is examined. Each of the seven chapters that constitute the BCP compliance index is evaluated based on the following four-point scale: (i) noncompliant; (ii) materially noncompliant; (iii) largely compliant; and (iv) compliant. We grade each chapter by assigning a numerical value to these ratings (from one for noncompliant to four for compliant). Given that the compliance index is based on a four-point scale, we thus arrive at a variable that ranges from 25 (lowest compliance) to 100 (highest compliance).

We further allow for factors that may influence the relationship between BCP compliance and bank stability by including two vectors, one for bank characteristics and one for country-level factors. *Bank _control*_{*ijt*-1} is a vector that captures bank portfolio characteristics. It includes measures of bank size proxied by the natural logarithm of total assets (*Inta*) and by the growth rate of total assets (*gta*). We note that the first of these variables may arguably increase or decrease bank stability and risk (Houston et al. 2010; Beck et al. 2013). The growth-rate measure reflects any expansion of the bank's balance sheet during the current year (compared to the previous year). Abedifar et al. (2013) employ this ratio as a proxy for bank growth and development strategies. As they expand and develop, banks might be further exposed to information asymmetry, since a considerable increase in activity may result in weaker screening standards and lower monitoring of investments. We also include the cost to income ratio (*cir*) to capture any cross-bank differences in efficiency, where higher values reflect managerial inadequacies (inefficiency) and thus a tendency for banks to take more risk (Abedifar et al. 2013; Beck et al. 2013). In addition, we use the ratio of noninterest income to total operating income

(*niiti*) to reflect a bank's business model and its tendency towards activity diversification. Finally, we use the ratio of liquid assets to deposits and short-term funding (*ladstf*) to assess sensitivity to bank runs, because banks with a higher proportion of liquid assets can be expected to experience lower bankruptcy costs, less information asymmetry, and a greater capacity to raise equity (Horváth et al. 2014).

Country _ *control*_{*jt*} is a vector of five macroeconomic and institutional variables commonly used in the stability literature (Houston et al. 2010; Demirgüç-Kunt and Detragiache 2011; Abedifar et al. 2013; Bitar and Tarazi 2019). It includes the GDP growth rate (*gdpg*) to reflect any potential cyclical behavior of the Basel regulation, the inflation rate (*inf*) to capture a country's general financial conditions, and the ratios of oil rent to GDP (*oil*), gas rent to GDP (*gas*), and mineral rent to GDP (*mineral*)⁶ as complementary measures of a country's economic independence, especially in relation to it being rich or poor in natural resources.

Finally, we employ the World Governance Index (*wgi*) as an additional measure to control for the quality of a country's political and institutional systems. Kaufmann et al. (2010) compute this index as the average of six governance dimensions (i.e., voice and accountability, political stability and absence of violence, government effectiveness, regulatory quality, rule of law, and control of corruption).

In our regression estimations, we winsorize all variables at the 1% and 99% levels to mitigate the effect of outliers. We follow Bitar and Tarazi (2019) and cluster at the bank level instead of the country level, for two reasons: First, our sample includes some countries with a much larger number of observations than others. Second, because we have 19 countries, clustering at a country level might create biased results (Nichols and Schaffer 2007).

3.3 Descriptive Statistics

Table 2 reports descriptive statistics for our samples of conventional and Islamic banks. Panels A and B present the mean, median, and standard deviation for the bank-level dependent and independent variables, while Panel C presents the summary statistics for the independent variables (i.e., the BCP compliance index and the seven chapters), and for the macroeconomic and institutional-environment control variables. Panel D presents the BCP compliance mean for each country and the corresponding year(s) of assessment.

In Panels A and B, we also report the outcomes of a Wilks' lambda test (λ) ,⁷ a Wilcoxon-Mann-Whitney test (Wilc), and a univariate analysis of variance test (F) for the equality of means for each financial ratio. The results of the statistical tests are presented in the last three columns of each panel and suggest that conventional banks are significantly different from Islamic banks with respect to all the financial ratios (except for the ratio of loan loss reserves to gross loans). The three tests demonstrate that the most significant difference between the two bank types relates to the measure of volatility of net-interest margins (SDNIM), with Islamic banks showing greater volatility. Finally, we note that the bank types differ significantly on our main

⁶ Oil, gas, and mineral rents mark the difference between the value of crude oil, gas, and mineral production at world prices and the total costs of production.

⁷ Wilks' lambda is the ratio of the within-group sum of squares to the total sum of squares. It takes on values between zero and one, with lower values indicating that the ratios are more capable of differentiating between conventional and Islamic banks.

Table 2General descriptive statistics.countries. Wilks' lambda (Wilks-\) is tthe ratios are more capable of differentifor the equality of means between Isla	al descriptive s' lambda (W ore capable o of means be	e statistics. Τ. 'ilks-λ) is the f differentiati tween Islami	his table pres ratio of the w ng between c c and conven	ents desc ithin-grou onventior tional bar	Table 2 General descriptive statistics. This table presents descriptive statistics for our variables using a sample of 651 countries. Wilks' lambda (Wilks-A) is the ratio of the within-group sum of squares to the total sum of squares. It takes on variable ratios are more capable of differentiating between conventional and Islamic banks. We also use a Wilcoxon-Mann-Whifor the equality of means between Islamic and conventional banks. See Table 13 in the appendix for variable definitions	ur variables usir he total sum of s We also use a V the appendix for	ug a sampl quares. It tr Vilcoxon-N variable d	e of 651 con akes on value Aann-Whitne efinitions	ventional ba s between ze y test (Wilc)	Table 2 General descriptive statistics. This table presents descriptive statistics for our variables using a sample of 651 conventional banks and 110 Islamic banks operating in 19 countries. Wilks' lambda (Wilks-M) is the ratio of the within-group sum of squares to the total sum of squares. It takes on values between zero and one, with lower values indicating that the ratios are more capable of differentiating between conventional and Islamic banks. We also use a Wilcoxon-Mann-Whitney test (Wilc) and a univariate analysis of variance test (F) for the equality of means between Islamic and conventional banks. See Table 13 in the appendix for variable definitions
Variables Co	Conventional banks (CBs)	banks (CBs)		Islam	Islamic banks (IBs)			Test statistics	S	
N	Mean	Median	ın SD	z	Mean	Median	SD	Wilks-A	Wilc	F-test
Panel A. Dependent variables	dent variabl	es								
Z-score (ln) 5(11 RGI 46	5031 3.61 4918 6.10	3.6 3.60	1.09 6 36	637 650	3.19 6.33	3.22 3.10	1.07	0.9844*** 0.9099	8.21*** 0.68	79.29*** 0.71
		0.71	1.76	672	1.77	0.75	2.96	0.9931***	-2.35**	39.36***
		4.70	9.58	457	7.53	3.70	9.75	0.9986^{**}	3.74***	6.00***
SDNIM 44	4449 0.60	0.35	0.95	651	1.36	0.57	2.18	0.9574^{***}	-10.98^{***}	227.01***
Panel B. Bank level control variables	level control	variables								
Inta 57				859	13.82	14.02	1.76	0.9962***	4.01^{***}	25.35***
gta 52	5273 15.97			754	25.27	18.27	36.99	0.9894^{***}	-7.82^{***}	64.51***
cir 55	5505 58.2	52.44	31.86	817	71.64	59.78	52.32	0.9872^{***}	-1.67^{***}	81.81***
niiti 55	5582 0.4	0.32		848	0.39	0.3	0.81	0.9998	2.81^{***}	0.01
ladstf 54	5419 45.64		42.03	786	58.22	29.42	70.71	0.9936^{***}	2.79***	40.07***
Panel C. Country level	ry level									
control variables										
BCP index 285										
		87.5	_							
Chapter 2 285										
Chapter 3 285		85	15.71							
Chapter 4 285			16.30							
Chapter 5 285	35 75.64		19.61							
Chapter 6 285			16.66							
Chapter 7 285	35 81.94	83	16.70							
wgi 285			0.65							
50		4.3	2.96							
		4.5	7.79							
oil 285		1.06	9.74							
gas 28	35 2.2	0.78	2.91							

Table 2 (continued)	ntinued)												
Variables	Conven	Conventional banks (CBs)	(CBs)		Islamic	Islamic banks (IBs)			Test statistics	S			
	z	Mean	Median SD	SD	z	Mean	Median	SD	Wilks-λ	Wilc	F-test		
mineral	285	0.35	0	0.81									
Panel D. BC	TP assess.	Panel D. BCP assessment across count	countries .	ries and years									
Country	N CBs	N CBs N obs. (%) N IBs	N IBs	N obs. (%)	Mean	Year BCP assessment	Country	N CBs	N obs. (%)	N IBs	N obs. (%)		Year BCP assessment
Albania	12	54.4	1	33.3	70.83	2005	akistan	28	30	8			2004
Bahrain	13	62.6	20	56	81.19	2005	saudi Arabia	8	100	4			2004, 2011
Bangladesh	32	88.1	7	94.3	49.76	2002, 2010	Singapore	22	36.4	1			2002, 2013
Egypt	31	71.4	ŝ	73.3	86.43	2002	South Africa		38	1			1999, 2010
Indonesia	81	65.1	10	37.3	70.16	2000, 2010	Syria	11	40	2			2008
Jordan	11	86.7	ŝ	73.3	77.50	2003	Tunisia		69.69	2			2001, 2012
Kenya	39	62	7	30	69.07	2003, 2010	Turkey		47.6	4			1999, 2011
Kuwait	9	83.3	7	51.4	73.81		UAE	19	78.2 9	6	53.3	90.71	2001
Lebanon	53	52.3	4	30	89.82	2001	UK		52	4		94.22	2002, 2011
Malaysia	35	73.5	18	49.2	91.73	2012							
Puto ** ***	* denote	s sionificance	v at 10% 50	*** ** and * denote significance at 1% 5% and 10% respectively	rechecti	velv							

***, **, and * denote significance at 1%, 5%, and 10%, respectively

 ${\ensuremath{\underline{\circ}}}$ Springer

dependent variable (the Z-score) with Islamic banks exhibiting a lower degree of stability; the difference is statistically significant in all three tests.

The mean of the BCP compliance index is 84.95% (Panel C), a much higher value than that reported by Demirgüç-Kunt and Detragiache (2011) and Ayadi et al. (2016). The high percentage is likely being driven by the inclusion of a large set of banks from emerging and developing countries. Ayadi et al. (2016) argue that the BCP index is much lower in the United States and in other developed countries than in developing countries. The high mean BCP index in this study suggests that banks in developing countries are moving toward global financial convergence through their compliance with the BCPs and international regulations. Finally, Panel D presents the number of conventional and Islamic banks in each country. For conventional banks, the sample is dominated by banks from the United Kingdom, and for Islamic banks, Bahrain contributes the largest number. We also note that for our sample period, the number of available observations (N obs. percent) is higher for conventional (58.4%) than for Islamic banks (52.1%). In other words, Bankscope provides information for conventional banks in approximately 8.8 out of our 15 sample years, while the coverage for Islamic banks is 7.8 out of 15 years.

4 Empirical Results

In Table 3, Panel A, we present regression results examining the effect of the BCP index on bank stability using Equations 1 and 2 after controlling for bank and country-level factors. Models 1–4 report the results for conventional banks, Models 5–8 report the results for Islamic banks, and Models 9–12 report the results for the full sample. We also present the results for the three components of the Z-score for each sample. These components comprise the ratio of return on average assets (ROAA), the standard deviation of ROAA (SDROAA), and the ratio of total equity to total assets (TETA). The Wald χ^2 (Chi-square) tests are highly significant for all models, and the R² values are relatively high, suggesting that the models are representative and that GLS random-effect regression is indeed an appropriate analysis technique.

We find that the BCP compliance index has a positive and significant effect on the natural logarithm of the Z-score of conventional banks (at the 1% level), Islamic banks (at the 5% level), and the full sample (at the 1% level). Economically, the estimated coefficients on BCP compliance in Models 1, 5, and 9 vary between 0.015 and 0.017, indicating that a one-unit increase in the compliance index is associated with an increase of nearly two percentage points in the natural logarithm of the Zscore. In contrast to Demirgüç-Kunt and Detragiache (2011), our results indicate that the Z-score is higher for conventional and Islamic banks in countries with higher BCP compliance, suggesting that these countries have sounder institutional settings. Demirgüç-Kunt and Detragiache (2011) and Ayadi et al. (2016) used a larger and more heterogeneous set of banks in countries with different regulatory regimes and different macroeconomic and institutional conditions; these factors could explain their failure to detect an effect of BCP compliance. This study focuses mainly on countries in which the banks all operate within similar financial, economic, and institutional conditions. In addition, our sample considers banks primarily drawn from developing countries, whereas Demirgüç-Kunt and Detragiache (2011) and Ayadi et al. (2016) employ samples that are dominated by banks operating in developed countries.

banks, Islamic banks, and the full sample using random effect GLS regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main independent variable is BCP compliance. We also use the components of the Z-score (SDROAA, ROAA, and TETA) as dependent variables. Panel B examines the impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score and its components. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	Table 3 BCP compliance and bank stability: Islamic vs. conventional banks Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional
independent variable is BCP compliance. We also use the components of the Z-score (SDROÁA, ROAA, and TETA) as dependent variables. Panel B examines the impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score and its components. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	banks, Islamic banks, and the full sample using random effect GLS regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main
compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score and its components. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	independent variable is BCP compliance. We also use the components of the Z-score (SDROAA, ROAA, and TETA) as dependent variables. Panel B examines the impact of BCP
parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score and its components. Standard errors are clustered at the bank level and are reported in
	parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions

stability
bank
uo
compliance
of BCP
impact
The
Ä

Variable	Conventional banks	al banks			Islamic banks	s			Full sample			
	Z-score	Components	Components of Z-score		Z-score	Components of Z-score	of Z-score		Z-score	Components	Components of Z-score	
Model #	(1)	SDROAA (2)	ROAA (3)	TETA (4)	(5)	SDROAA (6)	ROAA (7)	TETA (8)	(6)	SDROAA (10)	ROAA (11)	TETA (12)
BCP ($\alpha_{\rm BCP}$)	0.015***		-0.015***	0.085***	0.017**	-0.018			0.017***			0.097***
Inta	-0.042** -0.042**	-0.068***	-0.038 (0.034)	-2.998*** -2.998***	(0.009) 0.133 0.004)	-0.473***	-0.259	-7.585*** -7.585***	-0.035°	(cou.u) -0.067**	-0.094** -0.094**	-3.437
gta	-0.002***	0.000	-0.000	-0.014**	0.001	-0.005			-0.002***			-0.011*
cir	(0.001) -0.007***	(0.001) 0.01^{***}	(0.001) -0.012***	(0.006) 0.001	(0.001) -0.003 ***	(0.005) 0.005			(0.001) -0.006***			(0.006) 0.001
niiti	(0.001) -0.120	(0.002) 0.186	(0.003) 0.0334	(0.012) -1.253*	(0.001) -0.159	(0.005) 0.905			(0.001) -0.134			(0.011) -1.376**
lodef	(0.106)	0	(0.244)	(0.643) 0.047***	(0.220)	(0.679)			(660.0)			(0.612) 0.046***
lausu	(0.001) (0.001) 0.250***	0	(0.001)	0.04/ (0.013) 2.01 <i>5</i> ***	(0.001) (0.001)	(0.004) (0.004)			(0.001) (0.001)			(0.009) (0.009)
IBW	(0.051)	- U	(0.092)	(0.531)	0.156)	(0.270)			(0.048)			(0.561)
gdpg	0.04^{***} (0.009)	· · ·	0.086^{***} (0.017)	-0.01 (0.051)	0.02 (0.026)	0.04 (0.056)			0.038*** (0.008)			-0.015 (0.049)
inf	-0.026***	0.0	-0.026**	-0.036	0.019	-0.103***			-0.015 ***			-0.034
oil	-0.002	- 0	0.02***	0.12***	-0.013**	0.048***			-0.005			0.155***
gas	(0.005) -0.029*	~ ~	$(0.007) -0.041^{*}$	(0.034) -0.042	(0.005) 0.019	(0.013) 0.033			(0.004) -0.021			(0.032) 0.016
	(0.017)		(0.022)	(0.098)	(0.025)	(0.049)			(0.014)			(0.099)

Panel A. The impact of BCP compliance on bank stability	of BCP compl	liance on banl	k stability									
Variable	Conventional banks	al banks			Islamic banks	ks			Full sample			
	Z-score	Components of Z-score	of Z-score		Z-score	Components	Components of Z-score		Z-score	Components	Components of Z-score	
Model #	(1)	ROAA	ROAA (3)	TETA (4)	(5)	-SDROAA (6)	ROAA (7)	TETA (8)	(6)	-SDROAA (10)	ROAA (11)	TETA (12)
mineral	0.106***	-0.099***	0.054*	0.217*	0.028	-0.029	0.374*	-0.258	0.079***	-0.07***	0.02	0.126
Islamic	(770.0)	(070.0)	(050.0)	(611.0)	(0.047)	(con.n)	(777.0)	(coc.0)	(0.021) 0.204	(c20.0) -0.181	(/cu.u) 0.0414	(0.121) -3.293
$BCD \times Ielamic (\alpha, \beta)$									(0.640) -0.006	(0.894) 0.011	(2.435) 0.005	(6.646) 0.071
(Junter)									(0.008)	(0.011)	(0.029)	(0.090)
Constant	3.733***	1.244^{***}	3.024^{***}	50.67***	0.390	8.108^{***}	4.918	106.3^{***}	3.466***	1.51***	3.383***	55.98***
	(0.373)	(0.420)	(0.587)	(4.940)	(1.587)	(2.366)	(7.289)	(24.63)	(0.358)	(0.488)	(0.675)	(5.355)
Obs.	2559	2641	2709	2713	280	284	289	289	2886	2925	2998	3002
YFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.1398	0.187	0.2254	0.2872	0.3683	0.4187	0.471	0.4642	0.1432	0.1783	0.2149	0.303
χ^2 test (p value)	0.00***	0.00***	0.00***	0.00***	0.00***	0.00^{***}	0.00***	0.00***	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}
Panel B. Impact of BCP compliance ($\alpha_{BCP} + \alpha_{inter}$) on Islamic banks' stability (Model 9) and its components (Models 10 to 12)	JP complianc	e ($\alpha_{BCP} + \alpha_{int_{h}}$	$_{r}$) on Islamic ba	unks' stability	(Model 9) an	nd its compon	ents (Model	s 10 to 12)				
									0.11	0.001	-0.01	0.168^{**}
									(100.0)	(110.0)	((70.0)	(+00.0)

Table 3 (continued)

respectively
10%
and
5%,
1%,
at
significance
denote
*
and
*
**

To better understand what drives the positive association between BCP compliance and bank stability, we now focus on the three components of the Z-score. This allows us to examine which components are most responsive to the effects of BCP compliance: the return on average assets, the volatility of returns, or bank capitalization. In Table 3, Models 2 and 3 respectively show a negative impact of BCP compliance on the profits of conventional banks (at the 1% level) and on the volatility of returns (at the 10% level), while in Model 4, the association with capitalization is positive and significant (at the 1% level). For Islamic banks, the results are insignificant except in Model 8, where the association between BCP compliance and bank capital is positive (at the 10% level). The results for the full sample are similar to those for conventional banks, although the coefficient estimate for the return on average assets becomes insignificant. Finally, while our findings for the full sample in Model 9 suggest that BCP compliance has a positive effect on the stability of conventional banks ([α_{BCP}] is positive and significant), we do not find any significant impact of BCP compliance on the stability of Islamic banks ([$\alpha_{BCP}+\alpha_{inter}$], shown in Panel B, is not statistically significant). Note, however, that in Model 12, BCP compliance has a positive effect on the capital ratios of Islamic banks at the 5% level.

Together, the findings suggest that the positive association between BCP compliance and the Zscore of the two bank types is mainly due to the incentive to hold higher capital ratios in a strong regulatory environment that discourages excessive risk taking; this acts against the generation of higher profits and earnings volatility. Findings concerning the capital ratio are consistent with the recent literature on the role of institutional and regulatory factors as determinants of bank capital decisions. For instance, Bitar and Tarazi (2018) find that creditor protection can play a powerful role in incentivizing conventional banks to increase their capital ratios.

With regard to bank-level control variables, we find that bank size and Z-scores are negatively correlated, which is likely due to the negative effect of bank size on capital for both bank types (Abedifar et al. 2013; Beck et al. 2013). We also find that the growth in total assets is negatively associated with Z-scores, reflecting weaker screening standards and fewer monitoring incentives as assets grow. This trend is in line with the fact that regulatory authorities are more flexible with large banks in terms of capital requirements -a phenomenon that also explains the negative effect of the growth of total assets on bank capital. The cost to income ratio is negatively associated with bank Zscores, suggesting that managerial inadequacies reduce bank profitability and increase risk (Abedifar et al. 2013; Beck et al. 2013). The effect of bank-level control variables is less pronounced for Islamic banks, probably due to contradictory signs of the regression coefficient for different components of the Z-score. For instance, the liquidity ratios have a negative effect on bank profits and a positive effect on bank capital, resulting in an insignificant effect on the overall Z-score. With respect to the country-level control variables, we find that banks are more stable in countries with better GDP growth, higher mineral rents, lower gas rents, and lower inflation. When decomposing the Z-score, we observe that the positive effect of GDP growth is mainly driven by the ROAA (for both conventional banks and the full sample), while the negative effect of gas rents and inflation is driven by the ROAA for the sample of conventional banks.

5 Robustness Checks

5.1 BCP Index Components

To shed further light on the main results in Table 3, we now examine the impact of the seven chapters of BCP compliance on bank stability. While Demirgüç-Kunt and Detragiache (2011) and Ayadi et al.

(2016) examine the effect of the seven chapters in a single regression model, in this study, we separately introduce each chapter and examine its effect on bank stability, while taking account of the same bank- and country-level control variables mentioned above. In this way, we mitigate the effect of multicollinearity between the chapters on bank stability. For comparison purposes, we also report the effect of all chapters on bank Z-scores in a single regression model.

The results are presented in Table 4 where Panel A contains several important findings. First, the chapters examined in Models 2-7 all have a positive effect on conventional bank stability (at the 10% level or better). Chapter 2 on licencing and structure and Chapter 7 on cross-border banking generate the most pronounced effects on the Z-score of conventional banks, while the formal power of supervisors (Chapter 6) has the least pronounced effect. Overall, six of the seven chapters have a positive effect on the stability of conventional banks. For Islamic banks, we find positive and significant associations between the chapters incorporated in Models 10, 12, and 13 and the Z-score. Chapter 2 on licensing and structure is, again, the chapter that has the most pronounced effect on the Z-score, while Chapter 5 on information requirements generates the least pronounced effect. Overall, three of the seven chapters have a positive effect on the stability of Islamic banks. Second, if we compare the results with those obtained when all chapters are incorporated in a single model (see Models 8, 16, and 24), the findings become less pronounced for both conventional and Islamic banks and are similar to those reported by Demirgüc-Kunt and Detragiache (2011) and Ayadi et al. (2016). This supports our reservations regarding the problem of multicollinearity between different chapters if incorporated into a single model, which may explain the non-significant effect on bank stability reported in these studies.

Finally, the chapter results for the full sample (Islamic and conventional banks combined) resemble those reported separately for each type of bank. Specifically, our findings in Table 4, Models 17–23 continue to suggest that higher BCP compliance has a positive effect on the stability of conventional banks in Models 18–23 ([$\alpha_{BCP \ chapters}$] are positive and significant) and on the stability of Islamic banks in Models 18, 20, and 21 ([$\alpha_{BCP \ chapters} + \alpha_{inter}$], shown in Panel B, are positive and significant).

5.2 Subsamples

We examine the robustness of the previous results by exploring whether the relationship between BCP compliance and bank stability changes if we alter the sample composition to (i) exclude specific regions (namely the Gulf Cooperation Council [GCC], South East Asia [SEA], and the Middle East and North Africa [MENA]), (ii) exclude the United Kingdom, (iii) separately investigate listed and unlisted banks, and (iv) focus on specific periods of the economic cycle, i.e., the periods before (1999–2006), during (2007–09), and after (2010–13) the financial crisis. We also examine the effects of excluding groups of countries and banks, depending on their political stability, institutional environment, and efficiency scores.

The regional subsample results for the conventional, Islamic, and combined groups of banks are presented in Table 5, Panel A.1. We find that the association between BCP compliance and the Z-scores of conventional banks is significant and positive, and that this association is robust to the exclusion of banks in the GCC region, the SEA region, and the MENA region. For Islamic banks, the association between BCP compliance and Z-scores is marginally positive when excluding banks in the MENA region. However, the results become insignificant when excluding Islamic banks in the GCC and the SEA regions, suggesting that the positive association is mainly driven by those two regions. Finally, the results for the full sample do not show any significant impact of BCP compliance on the stability of Islamic banks ($[\alpha_{BCP}+\alpha_{inter}]$ in Panel A.2 is not statistically significant).

Table 4 BCP compliance and bank stability: Individual chapters. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional banks,
Islamic banks, and the full sample using random effect GLS regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main independent
variables are based on compliance with individual BCP chapters. These chapters include: Preconditions for effective banking supervision (chapter 1), licensing and structure (chapter 2),
prudential regulation and requirements (chapter 3), methods of ongoing supervision (chapter 4), information requirements (chapter 5), formal powers of supervisors (chapter 6), and
cross-border banking (chapter 7). Panel B explores the impact of BCP chapter compliance on the stability of Islamic banks using the natural logarithm of a bank's' Z-score. Standard
errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table A.2 in the appendix for variable definitions
Danal A. Tha invasor of individual RCD abavtore on bank ctability.

han!
on
ter
dual BCP chapt
vidua
t of individual
to
Daci
mpact
The im
Ε.
ane

	Conventional banks	banks							Islamic banks	ks						
Variable Model #	Z-score (1)	(2)	(3)	(4)	(2)	(9)	Ð	(8)	Z-score (9)	(10)	(II)	(12)	(13)	(14)	(15)	(16)
chapter 1 (α_{cp1}) chapter 2 (α_{cp2})	0.003 (0.003)	0.014***						-0.01* (0.006) 0.017**	0.011 (0.008)	0.025 ***						0.027 (0.019) 0.039*
chapter 3 (α_{cp3})		(0.002)	0.000***					(0.008) -0.002 (0.006)		(0.007)	0.01					(0.023) -0.009 (0.010)
chapter 4 (α_{cp4})			(700.0)	0.01***				-0.005			(000.0)	0.019***				0.025
chapter 5 (α_{cp5})				(700.0)	0.008***			0.006 0.006				(//////)	0.013 **			-0.042**
chapter 6 (α_{cp6})					(700:0)	0.003*		(c00.0) 0.005*					(0000)	0.001		-0.005
chapter 7 (α_{cp7})						(10000)	0.011***	(500.0) 0000-0-						(0000)	0.007	-0.002
Inta	-0.011	-0.023	-0.029		-0.02	-0.013	-0.031	(c.00.0) -0.041 **	0.163**	0.149*	0.171**	0.15	0.162**	0.159*	0.151*	0.125
	(0.019) 0.002***	(0.019) 0.002***	(0.019) 0.002***	(0.019)	(0.019)	(0.019)	(0.019) 0.002 ***	(0.019) 0.000***	(0.081)	(0.080)	(0.085)	(0.092)	(0.081)	(0.084)	(0.085)	(960.0)
gua	-0.005	(0.001)	(0.001)		(0.001)	(0.001)	(100.0)	-0.002	(0.001)	(0.001)	(0.001)	(100.0)	(100.0)	(0.001)	(0.001)	(0.002)
cir	-0.007***	-0.007***	-0.007***		-0.007***	-0.007***	-0.007***	-0.007 ***	-0.003***	-0.003***	-0.003***	-0.003**	-0.003***	-0.003***	-0.003***	-0.004**
niiti	(0.001) -0.029	(0.001) -0.05	(0.001) -0.03	(0.001)	(0.001) -0.046	(0.001) -0.035	(0.001) -0.028	(0.001) -0.095	(0.001) -0.220	(0.001)	(0.001)	(0.001) -0.334	(0.001) -0.287	(0.001) -0.326	(0.001) -0.243	(0.001)
	(0.101)	(0.103)	(0.102)		(0.102)	(0.100)	(0.102)	(0.104)	(0.233)	(0.232)	(0.215)	(0.245)	(0.224)	(0.247)	(0.239)	(0.233)
ladstf	0.001	0.001	0.001		0.001	0.001**	0.001	0.001	-0.000	-0.000	-0.000	-0.001	-0.000	-0.000	-0.000	-0.000
wei	0.255***	0.269***	(0.001) 0.256***		0.231***	0.255***	(0.001) 0.262***	0.291***	0.154	(100.0) 0.175	0.140	-0.0331	0.0472	(100.0)	0.150	0.331
b	(0.050)	(0.051)	(0.052)		(0.050)	(0.050)	(0.049)	(0.064)	(0.126)	(0.126)	(0.134)	(0.160)	(0.147)	(0.124)	(0.132)	(0.314)
gdpg	0.037***	0.04***	0.043***		0.038***	0.036***	0.039***	0.036***	0.034	0.034	0.03	0.023	0.033	0.032	0.028	0.036
	(0.008)	(0.008)	(0.008)		(0.008)	(0.00)	(0.008)	(0.009)	(0.024)	(0.023)	(0.023)	(0.023)	(0.023)	(0.023)	(0.024)	(0.025)
_	(0.004)	(0.004)	(0.004)		(0.004)	(0.005)	(0,004)	(0.004)	(0.010)	(0,010)	(110,0)	(0.010)	(010)	(0.010)	(0.010)	(0.012)
oil	-0.011***	-0.009***	-0.006		-0.01***	-0.012***	-0.011 ***	0.001	-0.019***	-0.014***	-0.02***	-0.02***	-0.017***	-0.02***	-0.017***	-0.017*
	(0.003)	(0.003)	(0.005)		(0.003)	(0.003)	(0.003)	(0.006) 0.006)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.004)	(0.005)	(600.0)
gas	-0.028*	-0.00/	-0.016)		-0.028*	-0.028*	-0.019	-0.016	0.025	0.028)	0.016	0.077	00.00	0.02	0.025	""/ 60.0 (0.046)
mineral	0.083***	0.089***	0.114***		0.074***	0.075***	0.095***	0.109***	-0.012	0.011	0.002	0.046	-0.015	-0.004	0.007	0.039
	(0.023)	(0.022)	(0.022)		(0.022)	(0.022)	(0.022)	(0.024)	(0.049)	(0.046)	(0.053)	(0.047)	(0.043)	(0.047)	(0.051)	(0.049)
Constant	4.260^{***}	3.496***	4.010^{***}		3.915***	4.313^{***}	3.936***	3.997***	0.585	-0.293	0.633	0.211	0.442	1.520	1.121	-0.430
	(0.350)	(0.362)	(0.332)	(0.353)	(0.320)	(0.315)	(0.333)	(0.429)	(1.538)	(1.369)	(1.484)	(1.517)	(1.419)	(1.384)	(1.400)	(1.583)
UDS. VEE	2,890 Vac	C1677	2,848 Vae	2,733 Vac	C/ 6,7	C16,2	2,8/3 Vac	2,000 Vac	542 Vac	000 Vac	529 Vac	501 Vac	000 Vac	Vac	542 Vae	790 Vac
1	0.1163	0.1397	0,1414	0.1285	0.1371	0.1202	0.1412	0,1460	0.3374	0.3441	0.3384	0.3488	0.3265	0.2976	0.3271	0.3937
of test (n-ve line)	+++000	+++00 0	******	*****	+++CC (******		+++000			****00 0	+++00 0	111100 0	*****	******	***00 0

Table 4 (continued)

Panel A. The impact of individual BCP chapters on bank stability

and M. The impact of mervicuar	Full sample	unit stubility						
'ariable Iodel #	Z-score (17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
slamic	-0.721	-0.621	-0.104	-0.258	-0.344	-0.095	0.258	-0.415
	(0.649)	(0.519)	(0.413)	(0.491)	(0.458)	(0.373)	(0.497)	(0.810)
hapter 1 (α_{cp1})	0.004							-0.007
slamic × Chapter 1 (α_{inter})	(0.003) 0.006							(0.006) 0.01
statile ~ Chapter 1 (uinter)	(0.008)							(0.014)
hapter 2 (α_{cp2})	()	0.015***						0.016**
		(0.002)						(0.008)
slamic × Chapter 2 (α_{inter})		0.005						0.017
		(0.007)						(0.011)
Thapter 3 (α_{cp3})			0.01*** (0.002)					0.001
slamic × Chapter 3 (α_{inter})			-0.003					(0.006) -0.024*
same Chapter 5 (Ginter)			(0.005)					(0.014)
hapter 4 (α_{cp4})				0.012***				-0.004
				(0.002)				(0.006)
slamic × Chapter 4 (α_{inter})				0.000				0.018
				(0.006)	0.009***			(0.014)
hapter 5 (α_{cp5})					(0.002)			0.003 (0.005)
slamic × Chapter 5 (α_{inter})					0.001			-0.013
anne Chapter 5 (Ginter)					(0.005)			(0.016)
hapter 6 (α_{cp6})					()	0.003**		0.003
						(0.001)		(0.003)
lamic × Chapter 6 (α_{inter})						-0.002		0.000
						(0.005)		(0.008)
hapter 7 (α_{cp7})							0.012*** (0.002)	0.002 (0.005)
slamic × Chapter 7 (α_{inter})							-0.007	-0.006
manne · chapter / (winter)							(0.006)	(0.013)
ta	-0.004	-0.016	-0.021	-0.019	-0.013	-0.005	-0.024	-0.037**
	(0.019)	(0.018)	(0.019)	(0.019)	(0.018)	(0.019)	(0.019)	(0.019)
a	-0.002***	-0.002***	-0.002***	-0.002**	-0.002***	-0.003***	-0.002***	-0.002**
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
r	-0.006*** (0.001)	-0.006*** (0.001)	-0.006*** (0.001)	-0.006*** (0.001)	-0.006*** (0.001)	-0.006*** (0.001)	-0.006*** (0.001)	-0.006*** (0.001)
iiti	-0.069	-0.101	-0.066	-0.181*	-0.094	-0.086	-0.072	-0.138
	(0.095)	(0.0962)	(0.095)	(0.102)	(0.095)	(0.094)	(0.095)	(0.099)
dstf	0.000	0.000	0.000	0.000	0.000	0.001	0.000	0.001
	(0.001)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
gi	0.242***	0.261***	0.242***	0.17***	0.209***	0.248***	0.25***	0.287***
	(0.047)	(0.047)	(0.048)	(0.049)	(0.048)	(0.047)	(0.047)	(0.061)
dpg	0.039***	0.042***	0.044***	0.038***	0.04***	0.0375***	0.039***	0.0379***
ıf	(0.008) -0.017***	(0.008) -0.015***	(0.008) -0.018***	(0.008) -0.014***	(0.008) -0.015***	(0.008) -0.016***	(0.008) -0.014***	(0.008) -0.018***
ш	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
1	-0.012***	-0.009***	-0.008**	-0.013***	-0.011***	-0.013***	-0.012***	-0.001
	(0.003)	(0.003)	(0.004)	(0.003)	(0.003)	(0.003)	(0.003)	(0.005)
15	-0.021	0.002	-0.013	-0.02	-0.025*	-0.022	-0.013	-0.001
	(0.014)	(0.014)	(0.014)	(0.014)	(0.014)	(0.013)	(0.013)	(0.017)
ineral	0.055**	0.07***	0.088***	0.077***	0.054**	0.055**	0.071***	0.085***
	(0.022)	(0.021)	(0.021)	(0.021)	(0.021)	(0.021)	(0.021)	(0.023)
onstant	4.014***	3.218***	3.794***	3.659***	3.694***	4.155***	3.695***	3.695***
bs.	(0.349) 3238	(0.357) 3325	(0.327) 3177	(0.347) 3034	(0.315) 3325	(0.310) 3225	(0.329) 3215	(0.418) 2886
FE	Yes							
2	0.1227	0.149	0.1439	0.1331	0.1435	0.1216	0.1472	0.1512
test (p-value)	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***	0.00***
anel B. Impact of BCP chapters (
r in the second second	0.01	eer z						0.003
	(0.007)							(0.015)
		0.021***						0.033***
		(0.006)						(0.011)
			0.007					-0.024*
			(0.005)					(0.013)
				0.012**				0.014
				(0.006)				(0.015)
					0.01**			-0.01
					(0.005)	0.001		(0.016)
					(0.005)	0.001		0.004
					(0.005)	0.001 (0.004)	0.005	

***, **, and * denote significance at 1%, 5%, and 10%, respectively.

The 167 conventional banks in the United Kingdom represent some 26% of the entire conventional bank sample. For this reason, we decided to exclude them from the analysis to examine the possibility that they might have caused a bias in the pattern of results. Table 5, Panel B.1, Models 1– 3 show that when all UK banks are excluded, the results are positive and significant for the two bank types. In addition, Panel B.2 suggests that the effect of BCP compliance on the stability of Islamic banks is positive and significant at the 10% level when excluding Islamic banks from the UK.

Table 5 BCP compliance and bank stability: Alternative samples. This table examines the impact of BCP compliance on the stability of conventional banks, Islamic banks, and the full sample across different subsamples using random effect GLS regressions. Panel A.1 provides a subsample analysis by excluding selected observations from the main sample: We first exclude the Gulf Cooperation Council (GCC) countries; second, we exclude the South East Asian (SEA) countries and finally we exclude the Middle East and North African (MENA)
countries. Panel B.1 breaks down the sample into three subsamples: First, we exclude banks from UK; second, we exclude unlisted banks; and finally, we exclude listed banks. Panel C.1 provides a sample break-down based on different time periods: First, we exclude the period before the 2007–2009 crisis; second, we exclude the 2007–2009 crisis period; and
finally, we exclude the period after the 2007–2009 crisis. In Panel D.1, we perform subsample analyses by considering three subsamples based on unstable political environments, weak protection of depositors, and highly efficient banks. In addition, Panels A.2, B.2, C.3, and D.4 examine the impact of BCP compliance on the stability of Islamic banks using the natural
logarithm of a bank's Z-scores across different subsamples. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions

	CBs	IBs	Full	CBs	IBs	Full	CBs	IBs	Full
Variable Model #	Z-score (1)	Z-score (2)	Z-score (3)	Z-score (4)	Z-score (5)	Z-score (6)	Z-score (7)	Z-score (8)	Z-score (9)
Panel A 1: Alternative samnles: Rr.		eakdown hv region							
		JCC region		Excluding the	excluding the SEA region		Excluding the MENA region	MENA region	
BCP (α_{BCP})	0.014^{***}	0.00	0.014^{***}	0.008*	0.001	0.009^{**}	0.014^{***}	0.016^{*}	0.016^{***}
	(0.003)	(0.010)	(0.003)	(0.004)	(0.010)	(0.004)	(0.003)	(0.00)	(0.003)
Inta	-0.048^{**}	0.091	-0.043**	-0.043*	0.245^{***}	-0.038*	-0.052^{***}	0.146	-0.049**
	(0.020)	(0.121)	(0.019)	(0.022)	(0.089)	(0.022)	(0.020)	(0.112)	(0.020)
gta	-0.002**	0.001	-0.002^{**}	-0.002	-0.000	-0.002	-0.002^{***}	0.001	-0.002***
	(0.001)	(0.002)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.001)
cir	-0.007***	-0.003^{***}	-0.006***	-0.008^{***}	-0.002	-0.006^{***}	-0.007^{***}	-0.003 ***	-0.006***
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
niiti	-0.0467	0.123	-0.052	-0.093	-0.076	-0.097	-0.09	-0.099	-0.125
	(0.104)	(0.436)	(0.098)	(0.155)	(0.227)	(0.142)	(0.114)	(0.228)	(0.109)
ladstf	0.001	0.000	0.001	0.001^{*}	0.000	0.000	0.0012^{*}	-0.000	0.001
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.008)	(0.001)	(0.001)
wgi	0.329^{***}	0.127	0.318^{***}	0.075	-0.025	0.067	0.286^{***}	0.151	0.275^{***}
	(0.069)	(0.328)	(0.066)	(0.060)	(0.179)	(0.057)	(0.053)	(0.164)	(0.049)
gdpg	0.026^{***}	0.005	0.025^{***}	0.035***	0.005	0.033^{***}	0.046^{***}	0.005	0.045^{***}
	(0.00)	(0.023)	(600.0)	(0.009)	(0.026)	(0.00)	(0.011)	(0.027)	(0.010)
inf	-0.029***	-0.018	-0.029***	-0.025^{***}	0.025^{**}	-0.015^{***}	-0.029^{***}	0.028^{**}	-0.017^{***}
	(0.005)	(0.034)	(0.005)	(0.005)	(0.012)	(0.005)	(0.004)	(0.012)	(0.005)
oil	0.042*	0.01	0.04*	-0.000	-0.016^{***}	-0.002	-0.001	-0.013^{**}	-0.004
	(0.022)	(0.075)	(0.020)	(0.005)	(0.006)	(0.004)	(0.005)	(0.006)	(0.004)
gas	-0.06^{**}	0.079	-0.05*	-0.013	0.019	-0.007	-0.039^{**}	0.01	-0.035^{**}

Table 5 (continued)									
Variable Model #	CBs Z-score (1)	IBs Z-score (2)	Full Z-score (3)	CBs Z-score (4)	IBs Z-score (5)	Full Z-score (6)	CBs Z-score (7)	IBs Z-score (8)	Full Z-score (9)
mineral	(0.028) 0.089^{***}	(0.069) 0.028	(0.026) 0.077***	(0.019) 0.108***	(0.023) 0.051	(0.016) 0.083***	(0.018) 0.111***	(0.025) 0.165	(0.015) 0.093***
Islamic	(0.023)	(0.059)	(0.022) 0.450	(0.024)	(0.053)	(0.024) -0.253	(0.036)	(0.132)	(0.034) 0.318
BCP × Islamic (α_{inter})			(0.686) -0.008 (0.008)			(0.983) -0.001 (0.012)			(0.709) -0.007 (0.000)
Constant	3.936*** (0 381)	1.701	(0.000) 3.853*** (0.376)	4.272*** (0 473)	0.117	(0.012) 4.008*** (0.461)	3.924*** (0.384)	0.482	(0.002) 3.692*** (0.378)
Obs.	2369	172	2541	1756	178	1934	2190	246	2436
YFE R ²	Yes 0.1575	Yes 0.2635	Yes 0.1618	Yes 0.1064	Yes 0.4582	Yes 0.1138	Yes 0.1633	Yes 0.4152	Yes 0.1628
x ² test (p value)	0.00^{***}	0.00***	0.00***	0.00***	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00***
Panel A.2 Impact of BCP compliance ($\alpha_{BCP} + \alpha_{inter}$) on Islamic banks' stability using different sub-regions 0.005 (0.008)	CP compliance (c	$x_{BCP} + \alpha_{inter}$) on Isl	amic banks' stabi 0.005 (0.008)	lity using diffe	rent sub-regio	15 0.008 (0.011)			0.009 (0.008)
Panel B.1 Excluding UK, unlisted, and listed banks Excluding UK banks	IK, unlisted, and listed Excluding UK banks	listed banks anks		Excluding unlisted banks	listed banks		Excluding listed banks	d banks	
BCP ($\alpha_{\rm BCP}$)	0.015***	0.019**	0.017***	0.02***	0.03**	0.021***	0.017***	0.01	0.018***
Inta	-0.004 -0.005)	0.147	-0.004 0.004	-0.001 -0.001	0.185	0.01	-0.042	0.126	-0.041 -0.05)
gta	-0.002***	0.001	-0.002**	-0.003*	-0.000	-0.002	-0.003***	-0.000	-0.002***
cir	-0.006^{***}	-0.003 -0.003	-0.000 -0.000 -0.000	-0.007***	-0.006	-0.007***	-0.006 ***	-0.003	-0.006***
niiti	-0.07 (0.110)	-0.187 (0.218)	-0.125 (0.106)	(0.168)	-0.086 (0.343)	-0.239* (0.139)	-0.059 (0.158)	-0.771^{***} (0.209)	-0.119 (0.149)

101

Table 5 (continued)									
Variable Model #	CBs Z-score (1)	IBs Z-score (2)	Full Z-score (3)	CBs Z-score (4)	IBs Z-score (5)	Full Z-score (6)	CBs Z-score (7)	IBs Z-score (8)	Full Z-score (9)
ladstf	0.001	-0.001	-0.001	0.001	0.001	0.001	0.001	-0.001	0.000
wgi	(0.001) 0.201*** 0.062)	-0.018 -0.018 (0.187)	(0.001) 0.188*** 0.058)	(200.0) 0.049 (270.0)	(0.045 - 0.045)	0.07 0.07	(0.001) 0.374*** (0.108)	(100.0) -0.308 (0.314)	(0.001) 0.318*** (0.102)
gdpg	0.026***	0.015	0.031***	0.056***	0.015	0.061***	0.019	0.014	0.017
inf	-0.024*** -0.024***	0.018 0.018 (0.012)	-0.015*** -0.015***	-0.028***	0.028	-0.018	-0.021*** -0.08)	0.022 0.022 0.0206)	-0.01
oil	-0.003	-0.013 ** (0.005)	-0.006 (0.004)	0.001	-0.015^{**} (0.007)	-0.004 (0.003)	-0.094** (0.0405)	0.04(0.043)	-0.066** (0.032)
gas	-0.024 (0.017)	0.01 (0.026)	-0.019 (0.014)	-0.014 (0.022)	0.036 (0.046)	-0.005 (0.019)	0.093 (0.057)	-0.052 (0.058)	0.058 (0.047)
mineral	0.023)	0.018 (0.044)	0.072***	0.114^{***}	-0.012 (0.059)	0.094***	0.054 (0.047)	-0.03 (0.055)	0.029 (0.041)
Islamic	~	~	-0.008	~	~	-0.451	~	~	0.353 (0.935)
BCP × Islamic (α_{inter})			-0.003 (0.008)			0.003			-0.008 (0.012)
Constant	3.190*** (0.441)	0.244 (1.684)	3.088*** (0.432)	2.805*** (0.543)	-0.902 (2.463)	2.537*** (0.568)	3.646*** (0.530)	1.435 (2.270)	3.522*** (0.524)
Obs. Vee	1891 Var	261 Vac	2152 V25	932 Var	138 Var	1070 Var	1490 Vac	126 Vac	1616 V25
R ² 	0.145	0.348	0.144	0.218	0.386	0.202	0.132	0.403	0.131
A use (p value) 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	BCP compliance ($(\boldsymbol{\alpha}_{\mathbf{B}\mathbf{C}\mathbf{P}} + \alpha_{inter})$ on]	0.00 (slamic banks' stal 0.014* (0.008)	o.oo bility affer exclu	ding UK, unli	0.00 sted, and listed 0.024** (0.009)	banks		0.01 (0.011)

			;			;			;
Womblo	CBs 7 coomo	IBs 7 20000	Full 7 20000	CBs 7 20000	T correction	Full 7 50020	CBs 7 20000	IBs 7 coom	Full 7
variaute Model #	(1)	2-50016 (2)	2-score (3)	2-score (4)	(5)	(9)	(1)	2-50015 (8)	(6)
Panel C.1 Alternative samples: Breal Excluding the		kdown by crisis/non-crisis periods period before the 2007–2009 crisis	n-crisis periods 007–2009 crisis	Excluding th	Excluding the 2007–2009 crisis period	risis period	Excluding the	Excluding the period after the 2007–2009 crisis	007–2009 crisis
BCP (α_{BCP})	0.006*	0.016*	0.008**	0.021***	0.005	0.021***	0.012***	0.004	0.014 ***
Inta	(0.003) -0.018 (0.022)	(0.008) 0.139 (0.115)	(0.003) -0.009 (0.022)	(0.003) -0.051 ** (0.023)	(0.010) 0.223* (0.116)	(0.003) -0.042* (0.022)	(0.003) -0.047** (0.020)	(0.011) 0.064 (0.096)	(0.003) -0.053*** (0.020)
gta	-0.002**	0.0001	-0.002*	-0.002*	-0.001	-0.002*	-0.002**	0.000	-0.002**
cir	-0.005 ***	-0.003	-0.005 ***	-0.007 ***	-0.003 **	-0.006 ***	-0.008***	-0.004*	-0.007 ***
niiti	0.08 0.08 0.178)	-0.198	(0.001) 0.002 (0.154)	-0.166	-0.061 -0.061	-0.181 -0.181 -0.113)	-0.15	-0.086	$-0.175^{(0.001)}$
ladstf	0.002**	0.000	0.001	0.001	0.002*	0.001	0.002***	-0.001	(201.0) 0.001 (100.0)
wgi	0.255***	0.073	0.252***	0.217***	0.146	0.219***	0.296***	0.146	0.266***
gdpg	0.041***	-0.033	0.032***	-0.007	0.053	-0.003 -0.013)	0.042***	0.017	0.041 ***
inf	-0.005	(0.022) 0.035*** (0.012)	0.007	-0.043*** -0.043***	(~c.0.0) -0.055***	-0.043*** -0.043***	-0.022*** -0.022***	0.022*	-0.013*** -0.013***
oil	(200.0) 0.001 (0.006	-0.018** (0.007)	(0.005)	(0.004) (0.004)	-0.005 -0.008)	(0.008** 0.008** (0.004)	-0.022*** -0.022*** (0.007)	(0.006)	(0.007) -0.019*** (0.005)
gas	-0.058*** (0.018)	0.012	-0.045 ***	-0.001	0.08**	0.006	-0.011	0.004	-0.011
mineral	0.085***	0.01	0.057**	0.125***	0.364	0.126***	0.092***	-0.000	0.067***
Islamic			-0.52			0.696			0.816 (0.872)
BCP × Islamic (α_{inter})			0.002 (0.008)			-0.01 (0.009)			-0.014 (0.011)

Table 5 (continued)									
Variable Model #	CBs Z-score (1)	IBs Z-score (2)	Full Z-score (3)	CBs Z-score (4)	IBs Z-score (5)	Full Z-score (6)	CBs Z-score (7)	IBs Z-score (8)	Full Z-score (9)
Constant Obs. YFE D2	3.807*** (0.388) 1396 Yes	0.753 (1.893) 178 Yes	3.575*** (0.386) 1574 Yes 0.1604	3.607*** (0.472) 1744 Yes	-0.146 (1.915) 171 Yes	3.426*** (0.458) 1915 Yes	3.796*** (0.431) 1735 Yes	1.700 (1.687) 171 Yes	3.611*** (0.432) 1906 Yes
χ^2 test (p value)	0.00***	0.00***	0.00^{***}	cc61.0 0.00***	1605.0	0.00***	/ 9010	0.00***	e/c1.0 0.00***
Panel C.2 Impact of BCP compliance	BCP compliance (a	$x_{BCP} + \alpha_{inter}$) on Isl	$(\alpha_{BCP} + \alpha_{inter})$ on Islamic banks' stability using different time periods 0.01 (0.008) (0.008)	ity using diffe	rent time perio	ds 0.011 (0.009)			0.001 (0.010)
Panel D.1 Other subsample considerations Unstable political er	ample considerations Unstable political environment	ns l environment		Weak protecti	Weak protection of depositors	S	Highly efficient banks	banks	
BCP (α_{BCP})	0.007 (0.006)	0.031 (0.026)	0.009 (0.006)	-0.21* (0.119)	-0.328** (0.159)	-0.296*** (0.098)	0.019^{***} (0.004)	0.015* (0.009)	0.025*** (0.003)
Inta	0.006 (0.046)	-0.115 (0.189)	0.001 (0.044)	0.163° (0.083)	-0.002 (0.219)	0.071 (0.068)	-0.043* (0.024)	0.154 (0.096)	-0.042^{*} (0.023)
gta	-0.001 (0.001)	0.005 (0.005)	-0.001 (0.001)	0.001 (0.003)	0.001 (0.003	0.001 (0.002)	-0.003^{***} (0.001)	0.001 (0.002)	-0.002^{**} (0.001)
cır niti	-0.005*** (0.001) -0.19	-0.012*** (0.004) 0.266	-0.006*** (0.001) -0.195	-0.007*** (0.002) -0.063	-0.002 (0.006) -0.147	-0.006*** (0.002) -0.213	-0.008*** (0.001) -0.197	-0.003*** (0.001) -0.261	-0.006*** (0.001) -0.222
ladstf	(0.141) 0.001	(0.555) -0.006	(0.142) 0.001	(0.389) 0.01	(0.334) -0.001	(0.254) -0.000	(0.185) 0.001	(0.233) -0.000	(0.159) (0.000)
wgi	(0.002) 0.806***	(0.006) -1.086	(0.002) 0.676**	(0.007) -1.187	(0.002) -2.402*	(0.001) -1.889*	(0.001) 0.183**	(0.001) 0.134	(0.001) 0.146^{**}
gdpg	(0.312) 0.035 (0.026)	(1.449) 0.051 (0.074)	(0.025) 0.031 (0.025)	(1.411) -0.032 (0.121)	(7.01) -0.195 (0.148)	(0c0.1) 0.066 (0.096)	(0.0/6) 0.014 (0.020)	(0.158) 0.016 (0.027)	(0.060) 0.023 (0.016)
inf	-0.031^{**} (0.013)	-0.07 (0.049)	-0.032*** (0.012)	-0.004 (0.012)	0.048* (0.027)	0.015 (0.012)	-0.044^{***} (0.007)	0.023* (0.012)	-0.018^{***} (0.006)

Table 5 (continued)									
Variable Model #	CBs Z-score (1)	IBs Z-score (2)	Full Z-score (3)	CBs Z-score (4)	IBs Z-score (5)	Full Z-score (6)	CBs Z-score (7)	IBs Z-score (8)	Full Z-score (9)
oil	0.01	0.127		-0.221*** (0.058)	-0.119	-0.195***		-0.016^{***}	-0.000
gas	-0.053 (0.039)	0.0247	-0.046 -0.035)	0.444**	0.301	0.426*** 0.096)	0.018	0.019 0.0260	0.016
mineral	-0.011	0.416*		0.346***	0.096	0.279***		0.046	0.046
Islamic	(6000)					-2.006 (1.560)			0.702
BCP × Islamic (α_{inter})			-0.003			0.025			-0.012 (0.008)
Constant	3.553*** (0.738)	2.805 (3.099)	3.529*** (0.710)	12.85** (5.898)	21.69** (9.702)	(5.337)	3.523*** (0.524)	0.493 (1.651)	2.957*** (0.469)
Obs.	842	95	937	177	76	253	1068	261	1329
	Yes 0.2626	Yes 0.4524	Yes 0.2657	Yes 0.3449	Yes 0.4659	Yes 0.3205	Yes 0.2098	Yes 0.3652	Yes 0.199
χ^2 test (p value)	0.00**	0.00**	0.00**	0.00**	0.00**	0.00**	0.00**	0.00**	0.00**
Panel D.2 Impact of BCP compliance ($\alpha_{BCP} + \alpha_{inter}$) on Islamic banks' stability using other subsamples 0.006 (0.009)	3CP compliance (c	$\chi_{ m BCP} + lpha_{inter})$ on Is	lamic banks' stabil 0.006 (0.009)	lity using other	· subsamples	-0.272*** (0.097)			0.013 (0.008)
***, **, and * denote significance at 1% , 5% , and 10% , respectively	significance at 1%,	5%, and 10%, resp	ectively						

Aside from regional and country effects, the association between BCP compliance and bank stability could be mainly confined to banks that are publicly listed, due to market discipline. Listing a bank on the market implies more stringent rules and stricter capital regulation and supervision, and thus less risky behavior. In Panel B.1, Models 4–9 present the results for subsamples of listed and unlisted banks. We find clear evidence that the effect of BCP compliance on a bank's Z-score is stronger when banks are publicly listed, particularly for Islamic banks. We hypothesize that, in contrast to unlisted Islamic banks, listed banks are more likely to seek international recognition through their compliance with BCP guidelines and by holding higher capital ratios. Therefore, listed Islamic banks are more prone to market discipline and regulatory pressure than unlisted ones, which could explain the strong positive association between BCP compliance and the Z-scores. Confirming the results in Panel B.1, Panel B.2 shows that the effect of BCP compliance on the stability of Islamic banks is positive and significant at the 5% level when excluding unlisted Islamic banks.

Table 5, Panel C.1 reports the results for sub-samples classified by the economic period. The findings provide clear evidence that the association between BCP compliance and conventional banks' stability is stronger for the period that preceded the financial crisis. In other words, the estimated coefficient on BCP compliance is smallest when the period before the financial crisis is excluded from the analysis and largest when the period of the crisis itself is excluded (exclusion of the post-crisis period produces an intermediate value for the coefficient). For Islamic banks, we observe a marginally positive effect only when excluding the period before the financial crisis. Overall, although the findings continue to support a positive effect of BCP compliance on the stability of conventional banks and a marginally positive effect of BCP compliance on the stability of Islamic banks, it appears that compliance does not preferentially increase bank stability in periods of economic distress. One reason for this is that our sample mainly covers banks in developing countries and these banks were less affected by the financial crisis than those in developed economies. Another reason is that some countries and regions in our sample are rich in natural resources and thus less exposed to economic turmoil than other countries (Bitar et al. 2016). Finally, the results for the full sample do not show any significant impact of BCP compliance on the stability of Islamic banks ($[\alpha_{BCP} + \alpha_{inter}]$ in Panel C.2 is not statistically significant).

We further check the robustness of our findings by studying whether the association between BCP compliance and bank Z-scores persists in countries with unstable political systems and weak institutional environments.⁸ In addition, we examine whether the positive effect of BCP compliance on bank stability persists for highly efficient banks.⁹ Table 5, Panel D.1 indicates that BCP compliance has a negative impact on the stability of conventional and Islamic banks in countries with weak protection of depositors and an insignificant effect in countries with less stable political institutions. However, the compliance effect becomes positive and significant once again for highly efficient banks. Panel D.2 shows that the effect of the BCP compliance index on the stability of Islamic banks is also negative in countries with weak protection of depositors and an engative in countries with weak protection of depositors banks is also negative in countries with weak protection of depositors banks is also negative in countries with weak protection of depositors. Overall, these results demonstrate that compliance with BCP is more effective for efficient banks in countries with better institutional environments and soundly based political systems.

⁸ We proxy for the stability of a country's political system by using an index of the durability of political institutions from the Political Regime Characteristics and Transitions of Polity IV database. We also proxy for institutional environments by using the creditor rights index from Djankov et al. (2007). We drop banks in countries with a durability index higher than the median. Likewise, we drop banks in countries with a creditor rights index that is higher than the median.

⁹ We proxy for bank efficiency by using bank efficiency scores estimated through data envelopment analysis (DEA). We drop banks with efficiency scores that are lower than the median.

Next, we estimate a series of quantile regressions to investigate whether the effect of BCP compliance on bank Z-scores varies in a significant way with different stability levels. An important feature of quantile regressions¹⁰ is that they allow for heterogeneous solutions that are conditioned on the level of the dependent variable.

Table 6 reports the results for the lower (Q25), the median (Q50), and the upper (Q75) quantiles of the Z-score distribution. The results in Panel A show that the estimated coefficients for the BCP compliance index are positive at all quantiles for the sample of conventional banks (Models 1–3) and for the full sample (Models 7–9), but not for the sample of Islamic banks. Moreover, while the coefficients on the Z-score increase somewhat across quantiles, the Wald tests do not detect any significant differences between the coefficients for the lower quantile and the upper quantile for either bank type, nor for the combined sample. As for Islamic banks in the full sample, Panel B suggests a marginally positive effect of the BCP index on stability for the lower quantile only.

5.4 Alternative Risk Measures

Our previous findings consistently show a positive and pronounced effect of BCP compliance on the Z-scores for conventional banks, and a positive but less pronounced effect for Islamic banks. We now focus on whether our findings persist when we re-estimate our regressions using alternative proxies for bank stability. First, we use three different measures of bank credit risk, namely the ratio of loan loss reserves to gross loans (LLRGL), the ratio of loan loss provision to total loans (LLPTL), and the ratio of nonperforming loans to gross loans (NPLGL). These three ratios measure loan quality, with higher values indicating poorer supervision and higher credit default risk (Beck et al. 2013; Abedifar et al. 2013; Bitar et al. 2016). Second, we use the standard deviation of net interest margins (SDNIM), with higher values indicating more volatile earning margins.

The results, presented in Table 7, Panel A, show clear evidence of a negative and significant association between the BCP compliance index and the three proxies of credit risk, as well as between compliance and the SDNIM. This holds true for the sample of conventional banks in Models 1–4 and the full sample in Models 9–12, while the results for Islamic banks are only significant for the SDNIM (Model 8). Economically speaking, the estimated coefficients for the BCP index in Models 1–3 vary between 0.012 and 0.059, indicating that a one-unit increase in the compliance index is associated with a decrease in credit risk that ranges between one percentage point (when using LLPTL) and nearly six percentage points (when using NPLGL). These results suggest that conventional banks in countries with higher BCP compliance have lower credit risk and are thus more stable.

To examine the robustness of our main finding, that BCP compliance is positively associated with the Z-score of conventional and Islamic banks, we further employ a battery of alternative estimation techniques. The results of these estimations are discussed in the following sections and confirm our key findings.

¹⁰ Quantile regression results are robust to outliers and distributions with heavy tails. In addition, quantile regressions avoid the restrictive assumption that error terms are identically distributed at all points of the conditional distribution.

Table 6 BCP compliance and bank stability: A quantile regression approach. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional banks, lslamic banks, and the full sample using conditional quantile regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main independent variable is BCP compliance. In Panel A, we present the 25th, 50th, and 75th quantiles of the dependent variable. Panel B explores the impact of BCP compliance on the stability of Islamic banks across the same quantiles. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	stability: A quan sample using con iance. In Panel A same quantiles. S	tile regression ap ditional quantile , we present the tandard errors ar	proach. Panel A regressions. Th. 25th, 50th, and 7 e clustered at the	examines the im e main depender 75th quantiles of e bank level and	pact of Basel Cc tt variable is the the dependent vi are reported in pi	lity: A quantile regression approach. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional le using conditional quantile regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main . In Panel A, we present the 25th, 50th, and 75th quantiles of the dependent variable. Panel B explores the impact of BCP compliance on the quantiles. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the	(P) compliance c n of a bank's Z explores the imp their coefficient	on the stability of -score (Z-score) act of BCP comp estimates. See T	conventional and the main bliance on the able 13 in the
	Conventional banks	banks		Islamic banks			Full sample		
Variable Model # Quantile	Z-score (1) Q25	Z-score (2) Q50	Z-score (3) Q75	Z-score (4) Q25	Z-score (5) Q50	Z-score (6) Q75	Z-score (7) Q25	Z-score (8) Q50	Z-score (9) Q75
Panel A. The impact of BCP compliance on bank stability: A comparison across quantiles	liance on bank st	ability: A compa	rison across qua	ntiles					
BCP (α_{BCP})	0.013***	0.017***	0.018***	0.018	0.013	0.016	0.017***	0.019^{***}	0.019***
Inta	-0.028	-0.047^{***}	-0.0487**	0.206	0.181^{*}	0.173	-0.025	-0.044**	-0.044**
	(0.022)	(0.018)	(0.022)	(0.152)	(0.106)	(0.150)	(0.023)	(0.019)	(0.020)
gta	-0.003^{***} (0.001)	-0.003** (0.001)	-0.004** (0.002)	0.002 (0.003)	-0.001 (0.003)	0.000 (0.002)	-0.002° (0.001)	-0.003	-0.003 (0.002)
cir	-0.009^{***}	-0.009***	-0.007***	-0.005***	-0.004***	-0.005***	-0.009***	-0.008***	-0.006***
	(0.001)	(0.001)	(0.002)	(0.0015)	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)
niti	-0.453**	-0.318**	-0.275*	-0.138	-0.26	0.063	-0.478** (0.101)	-0.297*	-0.323***
ladstf	0.002*	0.002**	0.001	0.001	0.001	000.0-	0.001	0.001 **	0.001
	(0.001)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.000)	(0.001)
wgi	0.138**	0.155***	0.092	0.014	0.083	0.017	0.113^{**}	0.138**	0.126**
gdpg	0.056***	0.047***	0.046***	0.036	0.024	0.006	0.056***	0.045***	0.048***
	(0.012)	(0.011)	(0.014)	(0.049)	(0.029)	(0.026) 0.025	(0.010)	(0.011)	(0.013)
ınt	-0.024*** (0.006)	-0.03*** 0.005)	-0.038*** 0.006)	0.016	0.03/***	0.027	-0.012** 0.006)	-0.021***	-0.027*
oil	0.002	-0.002	0.005	-0.007	-0.013**	-0.017**	0.000	-0.001	-0.000
	(0.005)	(0.003)	(0.010)	(0.00)	(0.006)	(0.008)	(0.003)	(0.003)	(0.007)
gas	-0.049^{***}	-0.024	0.013	-0.014	0.011	-0.013	-0.056^{***}	-0.02	0.003
minaral	(0.018)	(0.018)	(0.029)	(0.033)	(0.022) 0.077	(0.022) 0.015	(0.016)	(0.016)	(0.020)
111117101	(0.023)	(0.036)	(0.042)	(0.059)	(0.169)	(0.154)	(0.024)	(0.038)	(0.042)

Table 6 (continued)									
	Conventional banks	banks		Islamic banks	s		Full sample		
Variable Model # Quantile	Z-score (1) Q25	Z-score (2) Q50	Z-score (3) Q75	Z-score (4) Q25	Z-score (5) Q50	Z-score (6) Q75	Z-score (7) Q25	Z-score (8) Q50	Z-score (9) Q75
Islamic							-0.068	0.625	0.404
$\mathrm{BCP} imes \mathrm{Islamic}~(lpha_{\mathrm{BCP}_{\mathrm{inter}}})$							(2000) -0.001 (0000)	(0.494) -0.01 0.005	-0.010 -0.010
Constant	3.030^{***}	3.532***	3.951***	-0.429	0.449	1.325	(0.009) 2.626***	(0.000) 3.215***	(0.012) 3.691***
	(0.433)	(0.318)	(0.432)	(2.358)	(1.854)	(2.386)	(0.446)	(0.336)	(0.505)
Obs.		2606	2606	280	280	280	2886	2886	2886
YFE		Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2		0.148	0.128	0.347	0.355	0.358	0.14	0.148	0.137
χ^2 test (p value)		0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}
H ₀ : Q25 $\alpha_{BCPCBs} = Q75 \alpha_{BCPCBs}$							0.11		
H ₀ Q25 $\alpha_{\rm BCPIBs} = Q75 \alpha_{\rm BCPIBs}$				0.04			0.33		
H ₀ : Q25 $\alpha_{BCP_{inter}} = Q75 \alpha_{BCP_{inter}}$							0.39		
Panel B. Impact of BCP compliance on different quantiles of Islamic banks' stability ($\alpha_{BCP} + \alpha_{inter}$)	e on different qu	iantiles of Islami	c banks' stability	$\gamma (\alpha_{BCP} + \alpha_{inter})$				0000	0
							0.017*	0.009	0.009
							(0.008)	(0.006)	(0.011)

***, **, and * denote significance at 1%, 5%, and 10%, respectively.

Table 7 BCP compliance and alternative measures of risk. Panel A examines the impact of Basel Core Principles (BCP) compliance on the risk of conventional banks, Islamic banks, and the full sample using random effect GLS regressions. We use four alternative measures of bank risk: the ratio of loan loss reserves to gross loans (LLPTL), the ratio of nonperforming loans to gross loans (NPLGL), and the standard deviation of net interest margins (SDNIM). The main independent variable is BCP compliance. Panel B explores the impact of BCP compliance on the risk of Islamic banks using the same four risk measures. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	nce and altern ng random ef ns (LLPTL), i iance. Panel E in parenthese	ative measure fect GLS regruther ratio of no 3 explores the s below their of	s of risk. Pane sssions. We us nperforming l impact of BCI coefficient esti	A examines t the four alternat oans to gross oans to gross o compliance mates. See Ta	he impact of ive measures loans (NPLC on the risk of ble 13 in the	Basel Core F of bank risk JL), and the Islamic bank appendix for	rinciples (E : the ratio o standard de cs using the · variable de	SCP) compliant of loan loss n viation of n same four risement of n same four risement of the sam	unce on the ris serves to gro et interest ma sk measures.	sk of conventi ss loans (LLR rgins (SDNIN Standard error	onal banks, Is GL), the ratio (1). The main s are clustered	amic banks, of loan loss independent I at the bank
	Conventional	al banks			Islamic banks	ks			Full sample			
Variable Model #	LLRGL (1)	LLPTL (2)	NPLGL (3)	SDNIM (4)	LLRGL (5)	(6) (6)	(1) NPLGL	SDNIM (8)	LLRGL (9)	LLPTL (10)	NPLGL (11)	SDNIM (12)
Panel A. The impact of BCP compliance on bank risk	of BCP compl	iance on bank	risk									
BCP ($\alpha_{\rm BCP}$)	-0.049***	-0.012**	-0.059**	-0.009***	-0.048	0.012	-0.091	-0.024**	-0.05***	-0.014***	-0.058*	-0.012***
Inta	-0.653^{***}	(cnn.n) 0.004	(0.029) —1.114***	-0.045 ***	(0.008) -0.746	(/cu.u) -0.185	(0.122) -1.823	-0.083	-0.652^{***}	(cnn.n)	(0.029) -1.138***	(c.nu.) -0.044**
	(0.152)	(0.032)	(0.237)	(0.013)	(0.826)	(0.299)	(1.312)	(0.174)	(0.152)	(0.035)	(0.232)	(0.022)
gta	-0.022***	-0.003*	-0.029***	0.002**	-0.028***	-0.04	-0.025	-0.002	-0.023***	-0.003*	-0.03***	0.002**
cir	(0.004) 0.007	(0.001) -0.003	(0.007) 0.019	(0.001) 0.002^{**}	(0.009) 0.052^{***}	(0.005) 0.001	(0.019) 0.062	(0.005) -0.002	(0.004) 0.016^{**}	(0.001) -0.002	(0.007) 0.022*	(0.001) 0.001
	(0.007)	(0.003)	(0.012)	(0.001)	(0.006)	(0.006)	(0.043)	(0.002)	(0.007)	(0.002)	(0.012)	(0.001)
niiti	-0.0571	-0.309*	-0.0842	-0.174*	-1.514	-2.797^{**}	-3.199	-1.387*	-0.0693	-0.482^{**}	-0.0762	-0.245^{**}
	(0.529)	(0.184)	(1.034)	(0.101)	(1.642)	(1.123)	(2.911)	(0.746)	(0.516)	(0.195)	(1.015)	(0.114)
ladstf	0.012	-0.001	0.005	0.001**	0.011	-0.004	10.0	0.012*	0.011	-0.002	0.003	0.004**
wgi	0.594	-0.208**	0.756	0.029	0.678	(100.0) 869.0-	2.347	-0.058	0.518	-0.275 ***	0.728	-0.002
1	(0.380)	(0.091)	(0.584)	(0.041)	(1.207)	(0.502)	(1.760)	(0.219)	(0.364)	(0.101)	(0.546)	(0.047)
gdpg	-0.156***	-0.121***	-0.328***	-0.009	-0.342**	-0.027	-0.342	-0.035	-0.17***	-0.117***	-0.329***	-0.009
inf	(0.044) 0.006	(0.012) 0.023**	(0.067) 0.053	(0.008) 0.021***	(0.150) 0.087	(0.121)	(0.246) 0.11	(0.065) -0.035	(0.042) 0.002	(0.014)	(c90.0) 0.055	(0.009) 0.006
	(0.023)	(0.00)	(0.044)	(0.005)	(0.053)	(0.030)	(0.096)	(0.029)	(0.021)	(0000)	(0.040)	(0.009)
lio	-0.007	-0.007	-0.021	-0.007***	0.051	0.01	0.032	0.024	0.018	-0.004	-0.012	-0.001
	(0.021)	(0.005)	(0.025)	(0.002)	(0.044)	(0.011)	(0.073)	(0.016)	(0.019)	(0.005)	(0.025)	(0.004)
gas	0.555***	0.066**	0.024	-0.002	0.404**	0.099	0.332	0.19/**	0.528^{***}	0.0 /6**	0.05 (0.158)	0.038*
mineral	-0.433***	-0.084**	-0.463***	-0.032**	0.056	-0.229	-0.202	0.003	-0.385***	-0.09**	-0.464***	-0.029
	(0.126)	(0.042)	(0.175)	(0.014)	(0.291)	(0.182)	(0.232)	(0.070)	(0.122)	(0.038)	(0.164)	(0.019)
Islamic									0.432	-1.304	-1.316	-0.113

50	ß
-	3
-	2
-	b
2	
F	-

	Conventional	al banks			Islamic banks	ks			Full sample			
Variable Model #	LLRGL (1)	LLPTL (2)	NPLGL (3)	SDNIM (4)	LLRGL (5)	(9) (6)	(7) NPLGL	SDNIM (8)	LLRGL (9)	LLPTL (10)	NPLGL (11)	SDNIM (12)
BCP × Islamic (α_{inter})									(6.001) -0.004 0.073)	(2.665) 0.023 0.033	(7.696) 0.011 0.003)	(0.607) 0.008 0.007)
Constant	18.48***	2.571***	29.07*** (4.028)	1.770***	18.15	3.401	38.75* 01.86)	3.427	(C10.0) 17.97***	(cco.o) 2.821***	29.13*** 29.13***	1.876*** 1.876***
Obs.	2459	2449	(2000) (7000) (7000) (2000) (2000) (2000) (2000) (2000) (2000) (2000) (2000) (2000) (2000) (2000)	2636	246	248	164	283 283	2705	2697	2101	2919 2919
${ m YFE}$ R ²	Yes 0.1703	Y es 0.0862	Yes 0.1405	Yes 0.1533	Yes 0.4479	Yes 0.1942	Yes 0.3495	Y es 0.2837	Yes 0.1819	Y es 0.0832	Yes 0.1459	Yes 0.1207
χ^2 test (p value) 0.00*** 0	0.00***	0.00***	0.00***	0.00***	0.00^{***}	0.00***	0.00***	0.00^{***}	0.00^{***}	0.00^{***}	0.00***	0.00***
raiter D. IIIIpaci UI DC	л сошриансе		alliks 115K (WB	CP + Ctinter)					-0.053 (0.070)	0.009 (0.032)	-0.047 (0.088)	-0.003 (0.007)
2	ی	110 20 -1100	100	-								

***, **, and * denote significance at 1%, 5%, and 10%, respectively.

Table 8 BCP (to default meas Economic Inte supplementary the financial sti estimates. See	Table 8 BCP compliance and financial strength ratings. Panel A examines the impact of Basel Core Principles (BCP) compliance on various financial strength ratings and the distance to default measure for a sample of conventional banks, lalamic banks, and the full sample using random effect GLS regressions. We use three measures of financial strength ratings: the Economic Intelligence Unit's (EIU) credit risk index, the OECD's Country Risk Classification (CRC) index, and the International Country Risk Guide (ICRG) index. As a supplementary market measure, we use the distance to default (D-to-D). The main independent variable is BCP compliance. Panel B explores the impact of BCP compliance on the financial strength ratings and the distance to default measure of Islamic banks. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	ncial strength conventional 1 U) credit risk e use the dist e distance to endix for vari	thength ratings. Panel A ntional banks, Islamic b cdit risk index, the Of the distance to default the context of the the asture for variable definitions	I A examines banks, and OECD's Co alt (D-to-D). Ire of Islamic ns	the impact of Ba the full sample us untry Risk Class The main indep banks. Standard	sification (CF sification (CF endent varial errors are clu	cciples (BCP) effect GLS re RC) index, <i>z</i> ble is BCP c astered at the) complianc gressions. ⁷ und the Int ompliance. bank level	e on various fina We use three mea ernational Count Panel B explore and are reported	mcial strengtl asures of fina try Risk Gui es the impac in parenthes	h ratings and ncial strength de (ICRG) i t of BCP con es below thei	the distance ratings: the ndex. As a npliance on r coefficient
	Conventional banks	anks			Islamic banks				Full sample			
Variable Model #	EIU credit risk index (1)	CRC index (2)	ICRG index (3)	D-to-D (4)	EIU credit risk index (5)	CRC index (6)	ICRG index (7)	D-to-D (8)	EIU credit risk index (9)	CRC index (10)	ICRG index (11)	D-to-D (12)
Donol A Tho	Donal A. The immediate of DCD connellioned on contraction			tin co		ĥ						
BCP (α_{BCP})		-0.073***		-0.02***	-0.39***	-0.06***	0.16***	-0.011	-0.809***	-0.074***	0.28***	-0.019***
Inta	(ccu) -0.763***	(0.004) -0.066**	(0.014) 0.208	(86, cuu.u) -0.014	(0.081) -0.684	(0.012) -0.168**	(0.046) 0.112	(0.008) 0.017	(0.03) -0.773***	(0.004) -0.072***	(0.014) 0.215*	(cnn.n) 900.0-
	(0.228)	(0.028)	(0.133)	(0.027)	(0.616)	(0.084)	(0.485)	(0.083)	(0.218)	(0.027)	(0.127)	(0.025)
gta	-0.005	0.000	-0.002	-0.004*	-0.002	0.001	-0.002	-0.002	-0.005	0.001	-0.001	-0.004**
cir	(0.004) 0.009	(0.001) 0.001	(0.002) -0.004	(0.002) 0.004	(0.007) 0.001	(0.000)	(0.004) -0.003	(0.003) 0.001	(0.003) 0.008*	(0.000) 0.000	(0.002) -0.003	(0.002) -0.003
	(0.006)	(0.001)	(0.003)	(0.003)	(0.009)	(0.001)		(0.001)	(0.005)	(0.001)	(0.003)	(0.002)
niiti	0.158	-0.046	0.216	0.71	2.729*	0.011		0.383 **	0.257	-0.048	0.168	0.62
ladetf	(0.352)	(0.035) 0.000	(0.233)	(0.497) 0.000	(1.438)	(0.142) 0.001***		(0.169)	0.342)	(0.034) -0.001	(0.221)	(0.439)
TIGNET	(0.004)	(0.001)	(0.002)	(0.001)	(0.003)	(0000)		(0.001)	(0.003)	(0.001)	(0.002)	(0.001)
wgi	-7.243 * * *	-0.898***	4.634***	-0.218^{***}	-4.198^{**}	-0.65***		0.003	-7.032^{***}	-0.893***	4.792***	-0.11
-	(0.781)	(0.054)	(0.364) 0.275***	(0.084)	(1.849)	(0.200)		(0.219)	(0.724)	(0.053)	(0.342)	(0.071)
gupg	(0.052)	0.001	(01010)	(100.00)	-0.022	(0.014)	(0.068)	-0.012	0.047)	0.003)	0.018)	0.002
inf	0.022	0.021^{***}	-0.114^{***}	-0.021**	0.003	0.019^{***}	-0.061^{***}	-0.001	0.0186	0.02***	-0.107^{***}	-0.018^{**}
:	(0.018)	(0.002)	(6000)	(0000)	(0.044)	(0.004)	(0.023)	(0.019)	(0.016)	(0.002)	(0.008)	(0.008)
011	-0.104*** (0.035)	-0.011**	0.193*** (0.024)	-0.01***	-0.199	-0.023***	0.239*** (0.031)	-0.004	-0.106*** (0.028)		0.203*** (0.019)	-0.00/*** (0.002)
gas	0.952***	0.064***	-0.378***	-0.024	-0.076	0.004	-0.014	-0.024	0.819***	0.057***	-0.319***	-0.04***
mineral	(0.099) 0.092	(0.007) 0.025*	(0.044) 0.398^{***}	(0.018) 0.15^{***}	(0.188) 0.449	(0.021) 0.004	(0.123) -0.062	(0.018) 0.12	(0.085) 0.197	(0.007) 0.0273**	(0.041) 0.316^{***}	(0.012) 0.15^{***}
												-

Table 8 (continued)	(pa											
	Conventional banks	ınks			Islamic banks				Full sample			
Variable	EIU credit risk CRC indev indev	CRC indev	ICRG indev	D-to-D	EIU credit risk indev	CRC index	ICRG indev	D-to-D	EIU credit risk indev	CRC index	ICRG index	D-to-D
Model #	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)	(11)	(12)
	(0.160)	(0.0146)	(0.125)	(0.049)	(0.338)	(0.021)	(0.113)	(0.133)	(0.132)	(0.0132)		(0.046)
Islamic									1.69	-1.027	9.69***	-0.503
									(1.688)	(0.678)		(0.772)
$BCP \times Islamic$									0.442***	0.017*		0.003
(α_{inter})									(0.085)	(600.0)		(0.00)
Constant	117.5^{***}	8.995***	45.12***	4.673^{***}	81.74***	10.32^{***}	56.00^{***}	1.308*	117.9^{***}	9.201***		1.938^{***}
	(3.818)	(0.407)	(1.894)	(0.683)	(11.35)	(1.268)	(7.266)	(0.742)	(3.663)	(0.390)		(0.469)
Obs.	2882	2882	2882	1891	301	301	301	149	3183	3183		2040
YFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes		Yes
\mathbb{R}^2	0.457	0.392	0.393	0.475	0.596	0.557	0.684	0.487	0.46	0.4		0.476
χ^2 test (p value) 0.00***	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}	0.00^{***}		0.00^{***}
Panel B. Impact	Panel B. Impact of BCP compliance on		Islamic banks' risk ($\alpha_{BCP} + \alpha_{inter}$)	$(\alpha_{BCP} + \alpha_{inter})$	(,							
									-0.367^{***}	-0.057***	0.163^{***}	-0.016^{**}
									(0.076)	(0.00)	(0.027)	(0.008)

***, **, and * denote significance at 1%, 5%, and 10%, respectively.

I

5.5 Financial Strength Ratings and Distance-to-Default

First, we measure financial stability using financial strength ratings. We use three independent country risk assessment methods: the Economic Intelligence Unit's (EIU) credit risk index, the OECD's Country Risk Classification (CRC) index, and the International Country Risk Guide (ICRG) index. These measures are computed based on the opinion of a global network of economists and financial analysts who have access to quantitative and qualitative information about banks and their operating environments. As a result, they should provide a more accurate assessment of a bank's overall financial stability than accounting based indicators ¹¹ For the first two credit rating measures, higher values indicate higher credit default risk, while for the third credit rating measure, higher values indicate a lower level of exposure to credit risk.

We also use the Distance to Default (D-to-D) measure developed by Moody's KMV as a measure of default risk. This index is designed to assess the likelihood of financial distress and is based on the structural approach to calculating the Expected Default Frequency (EDF). According to Moody's, default occurs when the market value of a bank's assets (i.e., the value of the ongoing business) falls below its liabilities payable (the default point). See Table 13 for detailed definitions of the financial strength ratings and the D-to-D measure and for the associated data sources.

The findings reported in Table 8, Panel A, show that the BCP compliance index is positively associated with the financial strength ratings and negatively associated with the bank D-to-D measure. This holds for all models except for Model 8, which employs distance to default as a dependent variable for Islamic banks. In addition, the findings in Panel B show that BCP compliance has a less pronounced effect on the credit rating measures of Islamic banks, compared to conventional banks (in Models 9 to 11).

5.6 Other Estimation Techniques

In this subsection, we examine the robustness of the results using three alternative econometric specifications and standard error estimation methods. Table 9, Panel A provides the results of regressing the BCP index on banks' Z-scores. First, we use truncated regressions to address any biases related to the upper and lower tails of the distribution of observations for the dependent variable. In the second estimation, we use bootstrapped standard errors from 100 random resamples of the two bank types employed in the sample. In the third estimation, we correct for heteroscedasticity by using the White procedure. Importantly, the estimated coefficients of the BCP index are significant and positive in all models, except for Model 4 which applies truncated regressions to Islamic banks. The findings in Panel B further suggest that BCP compliance has a significant, positive effect on the Z-scores of Islamic banks at the 1% level (in Models 8 and 9).

5.7 Propensity Score Matching

Next, we employ the Propensity Score Matching (PSM) technique proposed by Rosenbaum and Rubin (1983) to verify the robustness of our results. PSM consists of matching bank observations based on the probability of increasing the country's BCP compliance index.

¹¹ Recent papers also used risk assessment provided by rating agencies as an alternative measure of financial stability. For instance, Vazquez and Federico (2015) use Moody's bank risk categories as an alternative risk measure to bank Z-score. Their findings show that credit ratings by rating agencies are important predictors of bank failure.

Table 9 Robustness checks: Alternative estimation techniques. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional banks, Islamic banks, and the full sample using three alternative estimation techniques and standard error calculations. First, we use truncated regressions to address any biases related to the upper and lower tails of the distribution of observations for the dependent variable. In the second estimation, we use bootstrapped standard errors from 100 random resamples of the two bank types employed in the sample. Finally, we correct for heteroscedasticity by using the White procedure. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main independent variable is BCP compliance. Panel B explores the impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score (Z-score. Finally, Panel C employs the propensity score matching technique proposed by Rosenbaum and Rubin (1983). Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	cs: Alternative estii using three alternation ion of observation aple. Finally, we c endent variable is l employs the proper coefficient estime	mation techniques tive estimation te s for the depende correct for heteros BCP compliance. nsity score matchi ates. See Table 13	. Panel A examir chniques and star nt variable. In th cedasticity by us Panel B explores ng technique pro in the appendix	tes the impact of E dard error calcula e second estimatic ing the White prov ing the impact of BC posed by Rosenba for variable defin	asel Core Princip tions. First, we us nh, we use bootstr nh, we use bootstr cedure. The main P compliance on turn and Rubin (1) itions	les (BCP) complia e truncated regress apped standard er dependent variabl the stability of Ísla 833. Standard err	nce on the stability ions to address any cors from 100 ranc e is the natural log mic banks using th ors are clustered at	estimation techniques. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional banks, Islamic emative estimation techniques and standard error calculations. First, we use truncated regressions to address any biases related to the upper and tions for the dependent variable. In the second estimation, we use bootstrapped standard errors from 100 random resamples of the two bank we correct for heteroscedasticity by using the White procedure. The main dependent variable is the natural logarithm of a bank's Z-score (Z- sis BCP compliance. Panel B explores the impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's opensity score matching technique proposed by Rosenbaum and Rubin (1983). Standard errors are clustered at the bank level and are reported timates. See Table 13 in the appendix for variable definitions	anks, Islamic the upper and the two bank s Z -score (Z - n of a bank's l are reported
Conventional banks	Islamic banks			Full sample			Conventional banks	banks	
Variable Model #	Truncated Z-score (1)	Bootstrap Z-score (2)	White Z-score (3)	Truncated Z-score (4)	Bootstrap Z-score (5)	White Z-score (6)	Truncated Z-score (7)	Bootstrap Z-score (8)	White Z-score (9)
Panel A: Alternative estimation techniques and BCP (α_{BCP}) 0.019***	on techniques and sta 0.019***	standards errors 0.015***	0.016***	0.005	0.016***	0.017***	0.018***	0.017***	0.018***
lnta	(0.003) -0.033**	(0.002) -0.039***	(0.002) -0.049***	(0.006) 0.143**	(0.004) 0.149**	(0.006) 0.16**	(0.003) -0.024*	(0.002) -0.035***	(0.002) -0.045***
gta	(0.016) -0.002*	(0.014) -0.002^{***}	(0.011) -0.003 ***	-0.000	(0.06/) 0.001	(c00.0) 0.001	(0.014) -0.001	(0.011) -0.002***	(0.011) -0.002***
cir	-0.007 ***	(0.001) -0.007***	(100.0)	-0.006 ***	-0.003 **	(0.001) -0.004**	-0.006 ***	(100.0)	-0.008 ***
niiti	(0.001) -0.645***	(0.001) -0.098	(0.001) -0.232^{***}	(0.001) -0.246	(0.001) -0.162	(0.001) -0.175	(0.001) -0.459***	(0.001) -0.134	(0.001) -0.256***
ladstf	(0.155) 0.002^{***}	(0.082) 0.001**	(0.088) 0.002^{***}	(0.248) 0.000	(0.239) -0.000	(0.205) 0.000	(0.146) 0.002^{***}	(0.085) 0.000	(0.084) 0.001**
wgi	(0.001) 0.071 (0.048)	(0.000) 0.267*** 0.0000)	(0.001) 0.119***	0.110	(100.0) 0.066	0.036	(0.001) 0.093**	(0.000) 0.248***	(0.000) 0.117***
gdpg	0.06***	0.038***	0.056***	(c11.0)	0.013	0.002	0.058***	0.038***	(cccu) 0.053***
inf	-0.02^{**}	-0.025 ***	-0.027***	0.021*	0.021*	0.025**	-0.018	-0.015***	-0.017**
oil	(0.008) 0.001	(0.004) -0.003	(0.004) -0.000	(0.013) -0.011 **	(0.013) -0.013 ***	(0.012) -0.012 ***	(0.008) 0.000	(0.004) -0.005**	(0.004) -0.002
gas	(0.004) -0.021	(0.004) -0.027** (0.011)	(0.003) -0.023 **	(0.005 -0.03 (0.020)	(0.004) 0.023 (0.024)	(0.004) 0.001 (0.017)	(0.003) -0.029**	(0.002) -0.021**	(0.002) -0.023***
mineral	0.084***	0.104^{***}	0.084***	0.037	0.032	0.068	0.095*** 0.076)	0.008***	0.072***
Islamic	(070.0)	(010.0)	((10.0)	(100.0)	(210:0)	(1000)	0.624	0.204	0.106
BCP × Islamic (α_{inter})							-0.01	(100.0)	(76670) -0.004

Table 9 (continued)									
Conventional banks	Islamic banks			Full sample			Conventional banks	anks	
Variable Model #	Truncated Z-score (1)	Bootstrap Z-score (2)	White Z-score (3)	Truncated Z-score (4)	Bootstrap Z-score (5)	White Z-score (6)	Truncated Z-score (7)	Bootstrap Z-score (8)	White Z-score (9)
Constant Obs.	3.304*** (0.324) 2113	3.679*** (0.275) 2606	3.448*** (0.237) 2606	1.827 (1.204) 235	0.390 (1.099) 280	-0.0593 (1.007) 280	(0.006) 3.137*** (0.296) 2355	(0.004) 3.466*** (0.225) 2886	(0.005) 3.052*** (0.248) 2886
YFE R ² 2 test (e tester)	Yes	Yes 0.1398	Yes 0.1514	Yes	Yes 0.3683	Yes 0.3836	Yes	Yes 0.1432	Yes 0.1533
X ⁻ test (p value) 0.00^{0} 0.00^{0} Panel B. Impact of BCP compliance on Islamic banks' stability ($\alpha_{BCP} + \alpha_{mer}$)	o.oo mpliance on Islamic t	o.00000000 banks' stability (α _{BCF}	0.00	0.00	0.00		0.008 0.006 0.006)	0.00^{+++} 0.011^{+++} (0.003)	0.00^{++++} 0.014^{++++} (0.005)
Panel C: Propensity score matching Con Trea	natching Conventional banks Treated/ controls	ıks Diff.	t stat	Islamic banks Treated/ controls	Diff.	t stat	Full sample Treated/ controls	Diff.	t stat
K-Nearest neighbors n = 10	3.737	0.169	1.68*	3.595	0.261	0.89	3.727	0.465	5.36***
n=15	3.737	0.288	3.00***	3.595 3.270	0.267	1.02	3.727	0.425	4.94***
n = 20	3.737 3.778 3.478	0.259	2.79***	3.595 3.310	0.276	1.12	3.727	0.424	5.08***
Kernel	3.737	0.123	1.2	Dropped			3.727	0.443	5.28***
Radius	3.737 3.476	0.261	6.65***	3.595 3.211	0.384	3.11***	3.727 3.454	0.273	7.28***
***, **, and * denote significance at	—	%, 5%, and 10%, respectively	sectively						

116

To implement PSM, we create a BCP compliance dummy variable that takes on a value of one if a country's BCP compliance index has a value greater than or equal to the median, and zero otherwise. We then estimate a logit model where we regress the BCP compliance dummy on all control variables in the baseline model and the year fixed effects. We use the estimated scores to produce matched observations between countries with higher and lower values of BCP compliance. Additionally, we employ three different matching methods: K-nearest neighbors (with n = 10, n = 15, and n = 20 nearest neighbors), Gaussian kernel matching, and radius matching. In the matched samples (presented in Table 9, Panel C), we continue to find evidence that conventional banks in countries with higher BCP compliance have higher Z-scores than matched conventional banks in countries with lower BCP compliance. We obtain very similar results for banks in the full sample, but the effect is less pronounced for the sample of Islamic banks. For each matching method, we report t statistics for the differences between the treated countries, with high BCP compliance, and the countries with low BCP compliance in the control group. For BCP compliance, the natural logarithm of the Z-score differences between the treated and control groups vary between 0.123 and 0.288% for conventional banks, between 0.123 and 0.276% for Islamic banks, and between 0.273 and 0.465% for the full sample. These differences are statistically significant at the 1% level in almost all models, except for the differences in the sample of Islamic banks, which are only significant when using the radius matching method.

5.8 Additional Control Variables

In this subsection, we address the issue of potential omitted variables. Thus, in addition to bank level, macroeconomic and institutional control variables, we include a series of four financial market regulation control variables to further check the robustness of the results.

Following Bitar and Tarazi (2019), we include the concentration ratio (*Concentration*), calculated as the ratio of the assets of the three largest banks to the total assets of the banking sector in the country.¹² Second, based on the World Bank's surveys on Banking Regulation and Supervision database,¹³ we use two market regulatory measures: market discipline and private monitoring (*Market discipline*) and entry requirements (*Bank entry*). The first measure captures the degree to which banks are required to disclose accurate information to the public and whether there are incentives to increase market discipline and private monitoring. The second measure controls for entry restrictions in terms of obtaining a banking license. Finally, we use the Heritage Foundation's financial freedom index (*Financial freedom*) to measure the extent of financial and capital market development as well as the openness to foreign competition (Bitar et al. 2018).

The findings are consistent with our previous results. As Table 10 shows, the explanatory variable (the BCP index) and the control variables maintain their signs and significance as before. As regards the financial market variables, we find that market discipline and private monitoring, entry requirements, and financial freedom have a positive and statistically significant effect on the stability of conventional banks, while the same effect appears to be absent for Islamic banks. In addition, concentration reduces bank stability. These findings suggest that the better the institutional

 $^{^{\}overline{12}}$ We also use C5 and recompute C3 and C5 according to the share of bank deposits. The results remain highly robust.

¹³ This information is collected based on surveys performed by the World Bank.

Table 10 BCP compliance and bank stability: additional control variables. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional banks. Islamic banks, and the full sample using random effect GLS regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main independent variable is BCP compliance. We address the issue of potential omitted control variables by including four measures to account for financial market conditions. These measures are: the concentration ratio (concentration), market discipline and private monitoring (market discipline), entry requirements (bank entry), and the financial freedom index (financial freedom). Panel B examines the impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score after controlling for potential omitted control variables. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	unce and bani und the full s BCP compl. entration ratic s. Standard (k stability: ad sample using 1 iance. We add o (concentrationes the impact errors are clus	lity: additional control variables. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional to using random effect GLS regressions. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main We address the issue of potential omitted control variables by including four measures to account for financial market conditions. These centration), market discipline and private monitoring (market discipline), entry requirements (bank entry), and the financial freedom index is impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score after controlling for potential are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable	ol variables. P: GLS regressi of potential si scipline and p bliance on the ank level and	anel A exam ons. The ma pointed cont rivate monit stability of l are reported	ines the imp in dependen rol variables oring (marke islamic bank in parenthes	act of Basel t variable is by includin et discipline) s using the r ies below thu	Core Principl the natural lc g four measu v, entry require natural logaritl eir coefficient	es (BCP) com garithm of a ves to account enents (bank hm of a bank estimates. Se	pliance on th bank's Z-sco for financial entry), and th s Z-score afth e Table 13 in	e stability of c re (<i>Z</i> -score) a market condi le financial fre er controlling the appendix	onventional nd the main tions. These edon index for potential for variable
	Conventional banks	al banks			Islamic banks	ıks			Full sample			
Variable Model #	Z-score (1)	Z-score (2)	Z-score (3)	Z-score (4)	Z-score (5)	Z-score (6)	Z-score (7)	Z-score (8)	Z-score (9)	Z-score (10)	Z-score (11)	Z-score (12)
Panel A. The impact of BCP compliance on bank	BCP compl	iance on bank		stability: Additional control variables	I variables							
BCP (α_{BCP})	0.012^{***}	0.021***		0.009***	0.014	0.026**	0.03**	0.017*	0.014^{***}	0.023***	0.015***	0.012***
Inta	(0.003) -0.041**	(0,003) -0.022	(0.004) -0.039*	(0.004) -0.044**	(0.011)	(0.012) 0.12	(0.012) 0.131	(0.009) 0.132	(0.003) -0.04**	(0.003) -0.023	(0.003) -0.034*	(0.003) -0.04**
	(0.020)	(0.027)	(0.020)	(0.020)	(0.103)	(0.110)	(0.101)	(0.00)	(0.020)	(0.026)	(0.020)	(0.019)
gta	-0.003***	-0.002*	-0.002**	-0.002**	0.001	0.001	0.001	0.000	-0.002***	-0.001	-0.002**	-0.002**
cir	(0.001) -0.007***	(0.001) -0.007***	(0.001) -0.007***	(0.001) -0.007***	(0.002) -0.003*	(0.002)	(0.001) 0.000	(0.001) -0.003***	(0.001) 0.007***	(0.001) -0.007***	(0.001) -0.006***	(0.001) -0.006***
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
niiti	-0.162	-0.155	-0.164	-0.106	-0.127	-0.129	-0.094	-0.159	-0.19^{**}	-0.18	-0.183*	-0.14
ى، 1-1-1	(0.101)	(0.134)	(0.116)	(0.107)	(0,232)	(0.221)	(0.201)	(0.219)	(0.095)	(0.118)	(0.105)	(0.100)
ladst1	(0.001)	0.001	0.001* (0.000)	0.001	0.000)	0.000)	0.001)	0.000)	0.001 (0.001)	-0.001)	-0.001	0.001
wgi	0.249***	0.233***	0.215***	0.284^{***}	0.065	-0.187	-0.039	0.085	0.229***	0.215***	0.197^{***}	0.263***
م مالمه م	(0.051)	(0.077)	(0.069)	(0.053)	(0.171)	(0.282)	(0.211)	(0.162)	(0.048)	(0.074) 0.045 ***	(0.068)	(0.050)
gupg	(0.008)	(0.013)	(110.0)	(0.009)	0.010	(0.023)	0.000	0.0260	(0.008)	(0.012)	(0.010)	(0.00)
inf	-0.024***		-0.015**	-0.025***	0.019	0.008	0.011	0.019	-0.016^{***}	-0.004	-0.006	-0.015^{***}
	(0.004)	(0.005)	(0.004)	(0.004)	(0.014)	(0.014)	(0.012)	(0.026)	(0.004)	(0.005)	(0.004)	(0.004)
oil	-0.004		0.005	-0.001	-0.013**	-0.005	-0.005	0.019	-0.005	-0.004	0.001	-0.004
gas	(0.029*)	-0.017	(con.o) -0.033**	(con.o) -0.022	(con.u) 0.02	0.019	(0.003 0.023	(2cu.u) 0.019	-0.022	-0.016	-0.025*	-0.019
0	(0.017)	(0.017)	(0.017)	(0.017)	(0.025)	(0.027)	(0.026)	(0.032)	(0.014)	(0.014)	(0.014)	(0.014)
mineral	0.092***	0.07^{***}	0.065***	0.114^{***}	0.04	0.052	0.066	0.028	0.072***	0.054^{**}	0.046^{**}	0.089^{**}

Table 10 (continued)												
	Conventional banks	al banks			Islamic banks	ıks			Full sample			
Variable Model #	Z-score (1)	Z-score (2)	Z-score (3)	Z-score (4)	Z-score (5)	Z-score (6)	Z-score (7)	Z-score (8)	Z-score (9)	Z-score (10)	Z-score (11)	Z-score (12)
Concentration	(0.023) -0.014*** (0.002)	(0.024)	(0.024)	(0.022)	(0.054) -0.007 (0.006)	(0.044)	(0.047)	(0.047)	(0.023) -0.014*** (0.000)	(0.023)	(0.023)	(0.022)
Market discipline	(700.0)	0.104^{**} (0.043)			(000.0)	0.088 (0.136)			(700.0)	0.089 ** (0.040)		
Bank entry			0.189**				-0.185				0.175**	
Financial frædom				0.005***				0.001			(21010)	0.005**
				(0.002)				(0.006)				(0.002) 0.002)
Islamic									0.013	0.112	0.238	0.038
BCP × Islamic (α_{inter})									-0.003	-0.005	-0.007	-0.004
									(600.0)	(600.0)	(0.00)	(0.002)
Constant	4.452***	3.597***	2.363***	0.005**	0.655	-0.999	0.594	0.559	4.261^{***}	3.31^{***}	2.139^{***}	3.646^{***}
	(0.389)	(0.514)	(0.601)		(1.785)	(2.096)	(2.010)	(1.626)	(0.387)	(0.497)	(0.579)	(0.374)
Obs.	2047	1525	2180		199	202	233	277	2246	1727	2413	2836
YFE	Yes	Yes	Yes		Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.151	0.173	0.171		0.373	0.402	0.406	0.375	0.151	0.179	0.178	0/143
χ^2 test (p value) 0.00*** 0.	0.00*** D compliance	0.00***	$00^{***} \qquad 0.00^{***} \qquad 0.00^{***}$	0.00***	0.00***	0.00***	0.00^{***}	0.00^{**}	0.00^{***}	0.00^{***}	0.00***	0.00^{***}
raiter D. IIIIpact of DC	er computation		alins staully	$(\alpha_{BCP} + \alpha_{inter})$	ŕ				0.011	0.010*	0000	0000
									0.008)	(0.009)	000.0)	0.007)

***, **, and * denote significance at 1%, 5%, and 10%, respectively

Table 11 BCP compliance and bank stability. Addressing endogeneity and selection biases. Panel A examines the impact of Basel Core Principles (BCP) compliance on the stability of conventional banks, lalamic banks, and the full sample using an instrumental variables (IV) approach to mitigate concerns about endogeneity and a Heckman estimation technique to mitigate concerns about a potential self-selection bias. For the IV approach, we use two estimation techniques: A two-stage least squares regression (2SLS) and a generalized method of moments (GMM) estimation. We use four instruments: The business regulation from the Fraser Institute's Economic Freedom of the World (EFW) Index, the legal origins, ethnic fractionalization, and independence. The main dependent variable is the natural logarithm of a bank's Z-score (Z-score) and the main independent variable is BCP compliance. Panel B explores the impact of BCP compliance on the stability of Islamic banks using the natural logarithm of a bank's Z-score. Standard errors are clustered at the bank level and are reported in parentheses below their coefficient estimates. See Table 13 in the appendix for variable definitions	ee and bank stability nic banks, and the 1 potential self-selecc iion. We use four i ppendence. The mai P compliance on the r coefficient estime	:: Addressing endc ùill sample using a tion bias. For the I nstruments: The b n dependent varial n e stability of Islam tes. See Table 13	genetity and selec in instrumental va V approach, we ui usiness regulation ble is the natural li nic banks using th in the appendix f	tion biases. Panel riables (IV) appro- se two estimation 1 from the Fraser ogarithm of a bani e natural logarithr or variable definit	A examines the i ach to mitigate c techniques: A tw Institute's Econc k's Z-score (Z-sc n of a bank's Z-s tions	mpact of Basel Cc concerns about enc o-stage least squan omic Freedom of ore) and the main ore. Standard err	re Principles (BCF logenetity and a Hd es regression (2SI the World (EFW) independent varial ors are clustered at	Compliance on the compliance on the communication extimation externant estimation. S) and a generalized index, the legal of ble is BCP compliance it the bank level and	e stability of technique to ed method of igins, ethnic nce. Panel B are reported
	Conventional b	al banks		Islamic banks			Full sample		
Variable Model #	2SLS Z-score (1)	GMM Z-score (2)	Heckman Z-score (3)	2SLS Z-score (4)	GMM Z-score (5)	Heckman Z-score (6)	2SLS Z-score (7)	GMM Z-score (8)	Heckman Z-score (9)
Panel A. The impact of BCP compliance	3CP compliance or	on bank stability							
BCP (α_{BCP})	0.025***	0.024***	0.017***	0.049***	0.053***	0.018**	0.034*** (0.006)	0.034***	0.019***
Inta	-0.057^{***}	-0.057	-0.05^{***}	(0.000) 0.117*	0.056	0.158*	-0.057^{***}	-0.057^{***}	-0.046**
	(0.012)	(0.012)	(0.018)	(0.060)	(0.171)	(0.095)	(0.012)	(0.012)	(0.018)
gta	-0.002^{***}	-0.002^{***}	-0.002^{***}	0.002	0.002	0.001	-0.001 **	-0.001^{**}	-0.002**
	(0.001)	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)	(0.001)	(0.001)	(0.001)
cir	-0.009***	-0.01^{***}	-0.008***	-0.005^{***}	-0.01^{***}	-0.004***	-0.01^{***}	-0.008^{***}	-0.01^{***}
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
nuu	-0.133*	-0.1.06* (0.0080)	961.0- (11.0)	CCU.0-	100.0	-0.106	**C/1.0-	-0.1 / /**	-0.17/*
ladstf	(0.002***	(0.002***	(0.114) 0.002^{***}	-0.000	-0.000	0.000	(0.001 **	(con.0)	0.001**
	(0.001)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)	(0.00)	(0000)
wgı	0.07/*	0.0/6*	د0.0 (۵.06)	0.17	0.209	0.047	0.108**	0.106**	0.050.00
gdpg	0.062***	0.061 ***	0.049***	0.031	0.029	-0.009	0.063***	0.063***	0.044***
	(0.011)	(0.011)	(0.011)	(0.029)	(0.031)	(0.026)	(0.011)	(0.011)	(0.010)
inf	-0.016^{***}	-0.016^{***}	-0.017^{***}	0.019	0.025	0.028^{**}	-0.007	-0.006	-0.007
	(0.005)	(0.005)	(0.005)	(0.012)	(0.021)	(0.012)	(0.004)	(0.004)	(0.005)
oil	0.004	0.004	0.002	-0.003	0.002	-0.011**	0.003	0.003	-0.000
	(0.003)	(0.003)	(0.004)	(0.005) 0.005)	(0.015)	(0.005)	(0.003)	(0.003)	(0.003) 0.003
gas	-0.03/** (0.016)	-0.03/** (0.016)	-0.027	-0.022 (0.019)	-0.014 (0.031)	0.006 (0.023)	-0.045^{***}	-0.045^{***} (0.013)	-0.026^{*} (0.014)

Table 11 (continued)									
	Conventional banks	anks		Islamic banks			Full sample		
Variable Model #	2SLS Z-score (1)	GMM Z-score (2)	Heckman Z-score (3)	2SLS Z-score (4)	GMM Z-score (5)	Heckman Z-score (6)	2SLS Z-score (7)	GMM Z-score (8)	Heckman Z-score (9)
mineral	0.073***	0.074***	0.069**	0.088	0.113	0.035	0.061***	0.062***	0.055**
Inverse Mills	(070.0)	(070.0)	(0.027) 0.046 0.040)	(/ c0.0)	(060.0)	(0.0/1) 0.383** /0.170)	(07070)	(070.0)	(0.026) 0.058 (0.041)
Islamic			(0+0.0)			(6/1.0)	0.756	0.752	0.006
							(0.499)	(0.499)	(0.662)
BCP × Islamic (α_{inter})							-0.012*	-0.011*	-0.002
							(0.006)	(0.006)	(0.008)
Constant	3.838***	3.832***	2.591^{***}	1.774	2.845	-2.429	3.694^{***}	3.687***	2.208***
	(0.286)	(0.286)	(0.412)	(1.128)	(3.162)	(1.474)	(0.278)	(0.278)	(0.401)
Obs.	2362	2362	2362	263	263	263	2625	2625	2625
YFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Adj. $\mathbb{R}^{2}/\mathbb{R}^{2}$	0.148	0.148	0.151	0.357	0.247	0.402	0.145	0.146	0.155
χ^2 test (p value)	0.00^{***}			0.00^{***}			0.00^{***}		
Sar/Han. J	0.264	0.254		0.167	0.147		0.307	0.288	
	(0.607)	(0.614)		(0.683)	(0.638)		(0.579)	(0.591)	
Panel B. Impact of BCP compliance on Islamic banks' stability $(\alpha_{n}, \dots, + \alpha_{n}, \dots)$	compliance on Isla	mic banks' stabilit	$(\frac{1}{M_{\rm D}} + \frac{1}{M_{\rm D}} + \frac{1}{M_{\rm D}})$						
							0.022^{***}	0.022^{***}	0.017^{**}
							(0.005)	(0.005)	(0.008)

environment (i.e., superior financial market regulation and increased competition), the more conducive the situation is to banking stability.

5.9 Addressing Endogeneity and Selection Biases

A country's BCP index could be endogenous, reflecting the impact of some omitted variables (e.g., market concentration, market regulation, financial freedom) or the impact of reverse causality. The findings in the previous sections demonstrate that the results are robust to the inclusion of additional control variables. Thus, we now address concerns about reverse causality. A common undertaking in the literature (e.g., Barth et al. 2013; Bitar and Tarazi 2019) is to address endogeneity by employing the Instrumental Variables (IV) approach. Thus, we use this approach and select four instruments chosen on the basis of the existing literature on law and finance (Beck et al. 2003; Barth et al. 2009; Barth et al. 2013; Öztekin 2015). These studies argue that legal origins and business regulation have important effects on banking regulation and are exogenous. However, a potential concern with using these two instruments is that they might affect bank capital decisions through channels other than the BCP index. We use two approaches to address this issue. Firstly, rather than employing legal origins and business regulation by themselves as instruments, we employ them along with two additional instruments: ethnic fractionalization and the (log) number of years that the country has been independent. The literature shows that these two instruments are less likely to have a direct impact on bank capital stability or some other omitted variable (Öztekin 2015). Additionally, while we instrument for the country's BCP index with legal origins and business regulation (along with ethnic fractionalization and independence), we control for the legal origins and business regulation in the second-stage regression.

The results are reported in Table 11, Models 1 and 2 for conventional banks, Models 4 and 5 for Islamic banks, and Models 7 and 8 for the full sample. We use two estimation techniques, a two-stage least squares regression (2SLS) and a generalized method of moments (GMM) estimation. The results provide clear evidence of a positive and significant association (at the 1% level) between the BCP compliance index and the Z-score in all models. For the full sample, the results in Panels A and B suggest that BCP compliance has a positive and significant effect on the stability of conventional banks (α_{BCP} is positive and significant). The same effect is observed in the case of Islamic banks ([$\alpha_{BCP}+\alpha_{inter}$] is positive and significant), but it is less pronounced.

In addition, we use a Heckman (1979) selection approach to mitigate concerns about a potential self-selection bias. The main objective of this technique is to correct for a self-selection bias based on whether or not countries are highly compliant with the Basel Core Principles. As a first step we define a dummy variable that takes on a value of one if the value of a country's BCP compliance index is greater than or equal to the median, and zero otherwise. We then estimate a probit model that regresses the dummy variable on the two instruments used before (i.e., rule of law and business regulation), as well as on the bank- and country-level control variables and the year-fixed effect from the baseline model. In the second-stage regression, we consider the natural logarithm of the Z-score as the dependent variable, the BCP compliance index as the independent variable (in addition to the same control variables), and the self-selection parameter (measured as the inverse Mills ratio) estimated from the first-stage regression. The findings of the second-stage regression (presented in Table 11, Panels A and B, Models 5, 10, and 15) continue to suggest that both conventional and Islamic banks are more stable in countries with a higher BCP compliance index.

6 Concluding Remarks

While previous studies that used the BCP index found no evidence of a significant association with bank stability and efficiency, this study suggests a positive effect of BCP compliance on the stability of banks in 19 countries. The findings are robust when the effects are examined separately for individual chapters of the BCPs. The association between compliance and stability is more pronounced for conventional than for Islamic banks. A deeper investigation into the components of the dependent variable (the Z-score) shows that the impact of compliance is confined mainly to the capital ratios of the two bank types.

Our findings have important implications from a regulatory perspective. Because compliance with the BCPs is effective in improving the stability of Islamic banks, it is likely that compliance with the CPFIRs can also positively affect Islamic bank stability, as they are modelled closely on the BCPs. Our findings persist across a battery of robustness checks that address potentially omitted variables, endogeneity concerns, and selection biases, and which also employ alternative estimation techniques. This first empirical assessment of Islamic banks shows that the BCPs are less effective at increasing the stability of Islamic banks than conventional banks. This difference leaves the open question of whether BCP standards should be amended to accommodate some of the specificities of Islamic banks, rather than having two separate sets of guidelines. One argument in support of a unified set of regulations is that it would lead to a more stable financial system in countries where the two bank types operate in tandem.

It is worth noting that the import of our results depend largely on the sample size, the choice of countries, and the validity of the accounting measures used as proxies for bank stability. Whilst increasing the sample size is beyond the potential of our study, since it would require additional surveys by the IMF and the World Bank, we attempt to overcome potential limitations related to measurement errors by using a wide range of proxies and econometric techniques. A next step in our analysis would be to explore the effect of the CPIFRs on the stability of Islamic banks and to compare it with the effect of the BCPs. In addition, it would be important to identify which BCP and CPIFR chapters are responsible for any significant effect on bank stability; in this context, the CPIFR chapters that reflect the specificities of Islamic banks will warrant particular attention. Unfortunately, data allowing for a comparative assessment of the two sets of guidelines are not currently available but will hopefully be incorporated into future research on this topic. Similarly, one could attempt to investigate whether the BCP and CPIFR guidelines have the same effect on Islamic bank efficiency by employing scores derived from nonparametric approaches or by using additional market-based data on stock returns and Tobin's Q, for instance. While the IFSB has asked banks to start reporting their data on the CPIFRs as of January 2016, the data are unlikely to be available before the end of 2018, which corresponds to a period outside the scope of this paper.

Acknowledgments We would like to thank participants at the 2018 Financial Engineering and Banking Society (FEBS), the MBF International Rome Conference on Money, Banking, and Finance and participants at the 34th French Finance Association Conference (Valence, France), Amine Tarazi (University of Limoges), Laurent Weill (University of Strasbourg), Anna Maripuu for collectiong data on financial strength ratings, Gabriela Maciel (RES), and Abdullah Haron (MCM) for their comments on an earlier version of our paper; Stephanie Fallas for editorial assistance; Jennifer Azar for administrative assistance. All errors and omissions are our own.

Appendix

Organization	IMF and World Bank Basel Core Principles	IFSB Core Principles for Islamic Finance
S-BuillZuiloli	(BCPs)	Regulation (CPIFR)
Program	Basel Core Financial Sector Assessment Program (FSAP)	Core Principles for Islamic Finance Regulation Working Group (CPIFRWG)
Starting date Objective	1999 To promote the stability and soundness of the financial sector, and to assess its potential contribution to growth and development.	January 2016 or later To provide a set of core principles for regulation and supervision, taking into consideration the specificities of Islamic banks and complementing the BCP compliance standards.
Principle 1	Objectives, independence, powers, and transparency	Retained unchanged
Principle 2	Permissible activities	Clear definition of licensed Islamic banks' permissible activities that are subject to supervision by regulatory authorities.
Principle 3	Licensing criteria	Retained unchanged
Principle 4	Transfer of significant ownership	Retained unchanged
Principle 5	Major acquisitions	The regulatory authority has the power to approve or reject, and impose more prudential requirements on major acquisitions, against prescribed criteria, including the establishment of cross-border operations, and to determine that corporate affiliations or structures do not expose the bank to undue risks or hinder ef- fective supervision.
Principle 6	Capital adequacy	Regulatory capital should be compliant with <i>Sharia'a</i> law. Accordingly, regulatory authorities require Islamic banks to adopt an appropriate capital adequacy approach by considering the particularities of Islamic banks (the extent of risk-sharing between bank shareholders (bank capital) and IAHs (depositors)).
Principle 7	Risk management process	Regulatory authorities require Islamic banks to establish a comprehensive risk management process, including an effective board of directors and senior management, appropriate steps to comply with <i>Sharia'a</i> law, and the development of contingency arrangements. This process depends on the Islamic banks' risk profile and their systemic importance.
Principle 8	Credit risk	Regulatory authorities require Islamic banks to create an adequate credit risk management process (taking into account a bank's risk appetite, its risk profile, and market and macroeconomic conditions) that covers the full credit lifecycle including credit underwriting, credit evaluation, and the management of Islamic banks' financing and investment
Principle 9	Problem assets, provisions and reserves	portfolios on a timely basis. Islamic banks should implement adequate policies to identify assets at risk early-on and to maintain an adequate amount of provisions and reserves.
Principle 10	Large exposure limits	Regulatory authorities determine whether Islamic banks have adequate policies to identify, measure, and control concentrations of

Table 12 Comparison between BCP chapters and CPIFR chapters

Table 12 (continued)

Principle 11	Exposures to related parties	risk. Regulators also set prudential limits to restrict bank exposures to single counterparties or groups of connected counterparties. To prevent the risk of conflict of interest with related parties, the supervisory authority requires Islamic banks to monitor transactions with these parties, to take appropriate steps to control or mitigate the risks, and to write off exposures in accordance with standard policies
Principle 12 Principle 13	Country and transfer risks Market risk	and processes. Retained unchanged Regulatory authorities determine whether Islamic banks have adequate market risk management (taking into account a bank's risk appetite, its risk profile, and market and macroeconomic conditions) to identify, measure, and control market risk on a timely
Principle 14	Liquidity risk	basis. Regulatory authorities provide the appropriate liquidity instruments for the needs of Islamic banks. These authorities also determine whether Islamic banks have adequate liquidity risk management (taking into account a bank's risk appetite, its risk profile, and market and macroeconomic conditions) to identify, measure, and control liquidity risk on a timely basis.
Principle 15	Operational risk	Regulatory authorities determine whether Islamic banks have an adequate operational risk management framework (taking into account a bank's risk appetite, its risk profile, and market and macroeconomic conditions) to identify, measure, and control operational risk on a timely basis.
Principle 16	Interest rates in the banking book	Rate of return risk instead of interest rates in the banking book. Regulatory authorities determine whether Islamic banks have an adequate system (taking into account a bank's risk appetite, its risk profile, and market and macroeconomic conditions) to identify, measure, and control rate of return risk on a timely basis. Regulators can also assess the capacity of an Islamic bank to manage the rate of return risk and any resultant displaced commercial risk, and to obtain sufficient information to assess the
Principle 17	Internal control and audit	behavior of IAHs and their maturity profiles. Regulatory authorities determine whether Islamic banks have adequate internal control frameworks to establish and maintain a properly controlled operating environment for the conduction of their business taking into account their risk profile.
Principle 18 Principle 19 Principle 20	Abuse of financial services Supervisory approach Supervisory techniques	Retained unchanged Retained unchanged Regulatory authorities employ adequate instruments to implement their supervisory approach taking into account the risk profile and systemic importance of an Islamic bank.
Principle 21	Supervisory reporting	

		The supervisory authority collects, reviews, and analyzes prudential reports and statistical returns from Islamic banks on both a solo and a consolidated basis, and independently verifies
Principle 22 Principle 23	Accounting and disclosure Corrective and remedial powers of supervisors	these reports through either on-site examina- tions or the use of external experts. Retained unchanged Regulatory authorities possess a range of tools to take corrective actions at an early stage to address unsafe practices or activities that could pose risks to an Islamic bank or to the banking
Principle 24	Consolidated supervision	system, i.e., the ability to revoke a banking license or to recommend its revocation. Regulatory authorities supervise the banking group on a consolidated basis; they adequately monitor and apply prudential standards to all aspects of the business conducted by the
Principle 25	Home-host relationships	banking group worldwide. Home and host regulatory authorities of cross-border banking groups share information and cooperate for effective supervision of the group and group entities. Supervisory authori- ties require the local operations of foreign Is- lamic banks to be conducted to the same
Principle 26	Not applicable	standards as those required of a domestic Is- lamic bank. Treatment of Investment Account Holders (IAHs). The regulatory authorities determine how IAHs are treated and also determine the various implications (including the regulatory
Principle 27	Not applicable	treatment, governance and disclosures, and capital adequacy and associated risk-absorbency features, etc.) relating to IAHs within their jurisdiction. <i>Sharia'a</i> governance framework. Regulatory authorities determine whether Islamic banks have a robust <i>Sharia'a</i> governance system to ensure an effective independent oversight of
Principle 28	Not applicable	Sharia'a compliance over various structures and processes within the organizational framework. The Sharia'a governance structure adopted by an institution offering Islamic financial services is commensurate and proportionate with the size, complexity, and nature of its business. The supervisory authority also determines the general approach to Sharia'a governance in its jurisdiction, and lays down key elements of the process. Equity investment risk. Regulatory authorities satisfy themselves through adequate policies and procedures including appropriate strategies, risk management and reporting processes for equity investment risk management, including <i>Mudarabah</i> and <i>Musharakah</i> investments in the banking book (i.e., financing on a profit-and-loss sharing basis), taking into ac- count Islamic banks' appetite and tolerance for risk. In addition, the supervisory authority

Table 12 (continued)

		ensures that Islamic banks: have in place ap- propriate and consistent valuation methodolo- gies; define and establish the exit strategies in respect of their equity investment activities; and have sufficient capital when engaging in equity
Principle 29 Not app	liashla	investment activities. Islamic "windows" operations. Supervisory
Finicipie 29 Not app	licable	authorities define what forms of Islamic "windows" are permitted in their jurisdictions.
		The supervisory authorities review Islamic
		windows' operations within their supervisory
		review process using the existing supervisory
		tools. The supervisory authorities in
		jurisdictions where windows are present satisfy
		themselves that the institutions offering such
		windows have the internal systems, procedures
		and controls to provide reasonable assurance
		that: (i) the transactions and dealings of the
		windows are in compliance with Sharia'a rules
		and principles; (ii) appropriate risk management
		policies and practices are followed; (iii) Islamic and non-Islamic business are properly segre-
		and non-Islamic business are properly segre- gated; and (iv) the institution provides adequate
		disclosures for its window operations.
		disclosures for his window operations.

Variable	Definition	Data sources
Z-score	A measure of bank insolvency calculated as the natural logarithm of ((ROAA+TETA)/SDROAA), where ROAA is the return on average assets, TETA represents the equity to assets ratio and SDROAA stands for the standard deviation of the return on average assets.	Authors' calculations
AROAA	A measure of risk-adjusted return on average assets. It is calculated as the return on average assets (ROAA) divided by the standard deviation of ROAA.	Authors' calculations
LLRGL	Bank reserves for loan losses divided by gross loans, multiplied by 100	Authors' calculations
LLPTL	Bank provisions for loan losses divided by total loans, multiplied by 100	Authors' calculations
NPLGL	Bank non-performing loans divided by gross loans, multiplied by 100	Authors' calculations
SDNIM	The standard deviation of the net interest margin for a three-year period	Authors' calculations
lnta gta	The natural logarithm of total assets The current year growth rate of bank total assets compared with the previous year's total assets.	Bankscope Bankscope
cir	The ratio of bank costs to bank income before provisions, multiplied by 100	Bankscope
niiti	Bank noninterest income to bank total operating income, multiplied by 100	Authors' calculations
ladstf	The ratio of liquid assets to deposits and short term funding. It measures and assesses the sensitivity to bank runs; therefore, it promotes financial soundness but it can also be interpreted as excess liquidity coverage.	Bankscope
BCP index	An overall index of BCP compliance. The index takes on values between 25 and 100, with higher values suggesting greater compliance with the BCPs.	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase
Chapter 1	This variable is the normalized sum of the rates of compliance with the sub-principles of principle 1 and measures the extent to which the preconditions for effective banking supervision have been met: 1(1): There should be clear responsibilities and objectives set by legislation for each supervisory agency; 1(2): Each supervisory agency should possess adequate resources to meet the objective set, provided that they do not undermine the autonomy, integrity, and independence of the supervisory agency; 1(3): There should be a suitable framework of banking laws setting bank minimum standards, including provisions related to the authorization of banking establishments and their supervisior; 1(4): The legal framework should provide the power to address concerns; 1(5): The legal framework should provide protection of supervisors for actions taken in good faith in the course of performing supervisory duties; and 1(6): There should be arrangements for interagency cooperation, including with foreign supervisors, for sharing information and protecting the confidentiality of such information.	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase

Table 13 Variable definitions

Variable	Definition	Data sources
	The variable takes on values between 25 and 100, with higher values indicating better adherence to these preconditions.	
Chapter 2	This variable is the normalized sum of the rates of compliance with principles 2–5; 2: Definition of permissible activities; 3: Right to set licensing criteria and reject applications for establishments that do not meet the standards set; 4: Authority to review and reject proposals for significant ownership changes; and 5: Authority to establish criteria for reviewing major acquisitions or investments. The variable takes on values between 25 and 100, with higher values indicating greater power of supervisors to approve a license and influence the structure of the new banking entity.	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase
Chapter 3	This variable measures the prudence and appropriateness of the minimum capital adequacy requirements that supervisors set. The variable is the normalized sum of the rates of compliance with principles 6–15: 6: Prudent and appropriate risk-adjusted capital adequacy ratios must be set; 7: Supervisors should evaluate banks' credit policies; 8: Banks should adhere to adequate loan evaluation and loan-loss provisioning policies; 9: Supervisors should set limits to restrict large exposures, and concentration in bank portfolios should be identifi- able; 10: Supervisors must have in place require- ments to mitigate the risks associated with related lending; 11: Policies must be in place to identify, monitor, and control country risks, and to maintain reserves against such risks; 12: Systems must be in place to accurately measure, monitor, and adequately control markets risks, and supervisors should have powers to impose limits or capital charge on such exposures; 13: Banks must have in place a compre- hensive risk management process to identify, measure, monitor, and control all other material risks and, if needed, hold capital against such risks; 14: Banks should have internal control and audit sys- tems in place; and 15: Adequate policies, practices, and procedures should be in place to promote high ethical and professional standards and prevent a bank being used by criminal elements. The variable takes on values between 25 and 100, with higher values indicating a greater compliance to the mini-	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase
Chapter 4	mum capital requirements. This variable measures the extent of ongoing supervision. The variable is calculated as the normalized sum of the rates of compliance with principles 16–20: 16: An effective supervisory system should consist of on-site and off-site super- vision; 17: Supervisors should have regular contact with bank management; 18: Supervisors must have the means of collecting, reviewing, and analyzing prudential reports and statistics from banks on a solo and consolidated basis; 19: Supervisors must have a means of independent validation of supervisory	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase

Table 13 (d	continued)
-------------	------------

Table 13 (co	ntinued)	
Variable	Definition	Data sources
	information, either through on-site examinations or the use of external auditors; and 20: Supervisors must have the ability to supervise banking groups on a consolidated basis. The variable takes on values between 25 and 100, with higher values suggesting higher levels of on-going supervision.	
Chapter 5	A measure of the required extent of a bank's internal financial records. This variable is the normalized rate of compliance with principle 21: Each bank must maintain adequate records that enable the supervisor to obtain a true and fair view of the financial condition of the bank, and must publish on a regular basis financial statements that fairly reflect its condition. The variable takes on values between 25 and 100, with higher values suggesting more stringent requirements for information disclosure.	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase
Chapter 6	A measure of the formal powers of supervisors, calculated as the normalized rate of compliance with principle 22: Adequate supervisory measures must be in place to bring about corrective action when banks fail to meet prudential requirements, when there are regulatory violations, or when depositors are threatened in any other way. This should include the ability to revoke the banking license or recommend its revocation. The variable takes on values between 25 and 100, with higher values indicating greater supervisory powers.	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase
Chapter 7	This variable measures the extent to which supervisors apply global consolidated supervision over internationally active banks. The variable is calculated as the normalized sum of the rates of compliance with principles 23–25: 23: Supervisors must practice global consolidated supervision over internationally active banks, and must adequately monitor and apply prudential norms to all aspects of the business conducted by these banks; 24: Consolidated supervision should include establishing contacts and information exchanges with the various supervisors involved, primarily the host country supervisory authorities; 25: Supervisors must require the local operations of foreign banks to be conducted with the same standards as those required of domestic institutions, and must have powers to share information needed by the home country supervisors of those banks. The variable takes on values between 25 and 100, with higher values suggesting a stronger movement towards	IMF/World Bank Basel Core Financial Sector Assessment Program (FSAP) da- tabase
wgi	global consolidated supervision. The world governance index is the average of six governance dimensions including: (1) voice and accountability, (2) political stability and absence of violence, (3) government effectiveness, (4) regula- tory quality, (5) rule of law, and (6) control of corruption (see Kaufmann et al. (2010) for details).	World governance indicators database (The World Bank and Kaufmann et al. (2010))
gdpg inf	Growth rate of GDP	World Development Indicators (WDI) World Development Indicators (WDI)

Variable	Definition	Data sources
	Inflation rate, based on changes in the consumer	
oil	price index Oil rents are the difference between the value of crude oil production at world prices and the total costs of production	World Development Indicators (WDI)
gas	costs of production. Natural gas rents are the difference between the value of natural gas production at world prices and the total costs of production.	World Development Indicators (WDI)
mineral	Mineral rents are the difference between the value of production for a stock of minerals (tin, gold, lead, zinc, copper, nickel, silver, bauxite, and phosphate) at world prices and the total costs of production.	World Development Indicators (WDI)
EIU credit risk index		The Economist – Economic Intelligence Unit
CRC index	The Country Risk Classification (CRC) index mea- sures the country credit risk and the likelihood that a country will repay its external debt. The index re- flects transfer and convertibility risk as well as cases of force majeure. It ranges between 0 and 7 with higher values indicating greater credit risk exposure.	www.OECD.org website - Country Risk Classification
ICRG index	The international Country Risk Guide (ICRG) index assesses three subcategories of risk: (1) political risk, (2) financial risk, and (3) economic risk. Used by banks and multinational corporations, the ICRG in- dex determines how financial, economic, and politi- cal risk might affect their business and investments now and in the future. The ICRG index ranges between 0 and 100 with higher values indicating less	www.prsgroup.com website – Interna- tional Country Risk Guide
D-to-D	risk exposure. The Moody's KMV distance to default is based on the Expected Default Frequency and is a measure of the probability that a bank will default over a specific period of time (typically one year). There are three key values that determine a bank's EDF credit measure: (1) the current market value of the bank (market value of assets), (2) the level of the bank's obligations (default point), and (3) the vulnerability of the market value to changes (assets volatility).	Moody's Analytics and authors' calculations
Concentration ratio Market discipline	Concentration of the three biggest banks in a country according to their share of total assets Market discipline and private monitoring is an indicator of disclosure of adequate information to the market. It ranges from 0 to 9 and is calculated by responding to the following 9 questions (1 if the answer is yes and 0 otherwise except for questions 8 and 9 where the opposite occurs): (1) Is subordinated debt allowed (or required) as capital? (2) Are	Authors' calculations Banking Regulation and Supervision database, World Bank, and authors' calculations

Table 13	(continued)
----------	-------------

Variable	Definition	Data sources
	financial institutions required to produce consolidat- ed accounts covering all bank and any non-bank financial subsidiaries? (3) Are off-balance-sheet items disclosed to the public? (4) Must banks dis- close their risk-management procedures? (5) Are directors legally liable for erroneous/misleading in- formation? (6) Do regulations require credit ratings for commercial banks? (7) Is an external audit by a certified/licensed auditor mandatory for banks? (8) Does accrued, unpaid interest/principal on non-performing loans appear on the income state- ment? (9) Is there an explicit deposit-insurance pro- tection system?	
Entry requirements	This variable is based on surveys by Barth et al. (2000, 2003, 2008, see details therein). The variable increases by 1 if the answer is yes to questions 1–8 of their survey, with no increase if the answer is no. The variable addresses 8 questions with higher values indicating stricter entry requirements: Regarding the legal submissions required for a banking license: (1) Is the legal submission require dor a banking license: (1) Is the legal submission require an intended organization chart? (3) Does the legal submission require first 3-year financial projections? (4) Does the legal submission require background/experience of future directors? (6) Does the legal submission require background/experience of future directors? (6) Does the legal submission require background/experience of future directors? (6) Does the legal submission require sources of funds in capitalization of the new bank? (8) Does the legal submission require information on the intended market differentiation of the new bank?	Banking Regulation and Supervision database, World Bank, and authors' calculations
Financial freedom index	The financial freedom index measures the efficiency	Heritage Foundation

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Abedifar P, Molyneux P, Tarazi A (2013) Risk in Islamic banking. Rev Financ 17:2035-2096
- Ayadi R, Ben Naceur S, Casu B, Quinn B (2016) Does Basel compliance matter for bank performance? J Financ Stab 23:15–32
- Barth JR, Caprio G, Levine R (2004) Bank regulation and supervision: what works best? J Financ Intermed 13: 205–248
- Barth JR, Caprio G, Levine R (2006) Rethinking bank regulation: till angels govern. Cambridge University Press, Cambridge, UK
- Barth JR, Caprio G, Levine R (2008) Bank regulations are changing: for better or worse? Comp Econ Stud 50: 537–563
- Barth J, Lin C, Lin P, Song FM (2009) Corruption in bank lending to firms: cross-country micro evidence on the beneficial role of competition and information sharing. J Financ Econ 91:361–388
- Barth J, Lin C, Ma Y, Seade J, Song F (2013) Do bank regulation, supervision and monitoring enhance or impede bank efficiency? J Bank Financ 37:2879–2892
- Beck T, Demirgüç-Kunt A, Merrouche O (2013) Islamic vs. conventional banking: business model, efficiency and stability. J Bank Financ 37:433–447
- Beck T, Demirgüç-Kunt A, Levine R (2003) Law, endowments, and finance. J Financ Econ 70:137-181
- Bitar M, Tarazi A (2019) Creditor rights and bank capital decisions: conventional vs. Islamic banking. Journal of Corporate Finance 55:69–104
- Bitar M, Hassan MK, Walker T (2017) Political systems and the financial soundness of Islamic banks. J Financ Stab 31:18–44
- Bitar M, Kabir MK, Hippler WJ (2018) The determinants of Islamic bank capital decisions. Emerg Mark Rev 35: 48–68
- Bitar M, Saad W, Benlemlih M (2016) Bank risk and performance in the MENA region: the importance of capital requirements. Econ Syst 40:398–421
- Carretta A, Farina V, Fiordelisi F, Schwizer P, Lopes F (2015) Don't stand so close to me: the role of supervisory style in banking stability. J Bank Financ 52:180–188
- Cihak M, Tieman A (2008) Quality of financial sector regulation and supervision around the world, IMF working paper 08/190. International Monetary Fund, Washington
- Demirgüç-Kunt A, Detragiache E (2011) Basel core principles and bank soundness: does compliance matter? J Financ Stab 7:179–190
- Demirgüç-Kunt A, Detragiache E, Tressel T (2008) Banking on the principles: compliance with Basel Core principles and bank soundness. J Financ Intermed 17:511–542
- Djankov S, McLiesh C, Shleifer A (2007) Private credit in 129 countries. J Financ Econ 84:299-329
- Doumpos M, Gaganis C, Pasiouras F (2015) Central bank independence, financial supervision structure and bank soundness: an empirical analysis around the crisis. J Bank Financ 61:S69–S83
- Gheeraert L (2014) Does Islamic finance spur banking sector development? J Econ Behav Organ 103:S4-S20
- Groeneveld H, de Vries B (2009) European co-operative banks: first lessons of the subprime crisis. Int J Co-Operative Manag 4:8–21
- Heckman JJ (1979) Sample selection bias as a specification error. Econometrica 47:153-161
- Horváth R, Seidler J, Weill L (2014) Bank capital and liquidity creation: granger causality evidence. J Financ Serv Res 45:341–361
- Houston JF, Lin C, Lin P, Ma Y (2010) Creditor rights, information sharing, and bank risk taking. J Financ Econ 96:485–512
- Islamic Financial Services Board (2015). Core principles for Islamic finance regulation (banking segment) (CPIFR)
- Kaufmann D, Kraay A, Mastruzzi M (2010) The worldwide governance indicators: methodology and analytical issues. Policy research working paper, no. WPS 5430. World Bank
- Khan F (2010) How Islamic is Islamic banking? J Econ Behav Organ 76:805-820
- Klomp J, de Haan J (2012) Banking risk and regulation: does one solution fit all? J Bank Financ 36:3197–3212
- Klomp J, de Haan J (2014) Bank regulation, the quality of institutions, and banking risk in emerging and developing countries: an empirical analysis. Emerging Market Financ Trade 50:19–40
- Leaven L, Levine R (2009) Bank governance, regulation and risk taking. J Financ Econ 93:259-275
- Mollah S, Zaman M (2015) Shari'ah supervision, corporate governance, and performance: conventional vs. Islamic banks. J Banking Finance 58:418–435
- Mollah S, Hassan K, Al-Farooque O, Mobarek A (2017) The governance, risk-taking, and performance of Islamic banks. J Financ Serv Res 51:195–219
- Nichols A, Schaffer ME (2007) Clustered standard errors in Stata. In: United Kingdom Stata users' group meetings 2007. Group, Stata Users

- Öztekin Ö (2015) Capital structure decisions around the world: which factors are reliably important? J Financ Quant Anal 50:301–323
- Podpiera R (2006) Does compliance with basel core principles bring any measurable benefits? IMF Staff Papers 53:306–326
- Rosenbaum PR, Rubin DB (1983) The central role of the propensity score in observational studies for causal effects. Biometrika 70:41–55
- Sundararajan V, Marston D, Basu R (2001) Financial system standards and financial stability the case of Basel Core principles, IMF working paper 01/62. International Monetary Fund, Washington
- Vazquez F, Federico P (2015) Bank funding structures and risk: Evidence from the global financial crisis. Journal of Banking & Finance 61, 1–14

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.